**UNIVERSITY OF KWAZULU-NATAL** 

**COLLEGE OF LAW AND MANAGEMENT STUDIES** 

SCHOOL OF LAW- HOWARD COLLEGE

Liability for Injuries to Seafarers- a critical analysis of the extent of shipowner liability under the Maritime Labour Convention.

MIKAELA JAUDEN JOSEPH

215002928

This mini-dissertation is submitted in partial fulfilment of the requirements for the degree of Master of Laws (Maritime Law)

Supervisor: Mrs Deepa Lamb

2022

# **ABSTRACT**

The main aim of this dissertation is to determine whether the Maritime Labour Convention has been successful in ensuring the fair and equal treatment of seafarers in relation to shipowner liability. This dissertation aims to do this by analysing the national legislation in three different jurisdictions (The United States of America, United Kingdom and South Africa) before the implementation of the Maritime Labour Convention. The national legislation in the respective jurisdictions after the implementation of the Maritime Labour Convention will then be analysed to determine whether or not the Convention has been successful in protecting the rights of injured seafarers and whether they are treated fairly and equally in all three of the respective jurisdictions. The dissertation will comprise of desktop-based research and will make use of comparative research methodology. This dissertation will comprise of five chapters. Chapter one introduces the problem topic and provides the background to the problem. Chapter two examines the MLC, it provides information on the aims of the MLC, how it was developed, the implementation of the MLC and the provisions for shipowner liability in respect of injured seafarers. Chapter 3 examines the history of the national laws in the respective three jurisdictions before the implementation of the MLC and analyses the extent to which an injured seafarers' rights were protected and where a shipowner's liability ends. Chapter 4 examines the national laws in the respective three jurisdictions after the implementation of the MLC and analyses the extent to which an injured seafarers' rights are now protected and examines exactly where a shipowner's liability ends. Chapter 5 concludes the dissertation. It restates the key research questions and the resulting conclusions. After analysing the national legislation in the abovementioned jurisdictions, it is evident that the domestic legislation protecting seafarers, even prior to the implementation of the MLC in these jurisdictions, already extensively protected seafarers employed on board vessels registered in the UK and SA. The MLC will however have a substantial effect on the ability of port authorities in member states to inspect and take action against owners of substandard vessels. The MLC is not above criticism and its shortfalls have become apparent in the crew change crisis during the Covid-19 pandemic. Nevertheless, the MLC is still a great step forward in the consolidation and harmonisation of compliance and inspection procedures followed by flag states and port authorities, thus increasing the protection and enforcement of seafarers' rights.

# **TABLE OF CONTENTS**

DECLARATION OF ORIGINALITY	5
ACKNOWLEDGEMENTS	6
LIST OF ABBREVIATIONS	7
CHAPTER ONE: INTRODUCTION	
1.1 Introduction	8
1.2 Scope and limitations of the study	11
1.3 Research Objectives and Rationale	12
1.4 Research methodology and limitations	17
1.5 Structure of the dissertation.	17
CHAPTER TWO: THE MARITIME LABOUR CONVENTION	
2.1 Introduction.	19
2.2 The primary aims of the Maritime Labour Convention	21
2.3 The basic structure of the MLC	22
2.4 How the MLC aims to protect more seafarers	23
2.5 Minimum standards for seafarers	24
Minimum Age	24
Seafarers' employment agreements	25
Wages	25
Repatriation	26
Compensation for the vessel's loss or foundering	27
Medical care on board and ashore	27
Shipowner Liability	28
2.6 How the MLC aims to protect shipowners	28
2.7 How the MLC is implemented	30
2.8 Advantages of the MLC	32
2.9 Criticism of the MLC	33

CHAPTER THREE: SEAFARER'S RIGHTS PRIOR TO THE	<u> MLC</u>
3.1 The United States of America	35
3.2 The United Kingdom	38
3.3 South Africa	41
CHAPTER 4: SEAFARER'S RIGHTS AFTER THE IMPL MLC	EMENTATION OF THE
4.1 The United States of America	45
4.2 The United Kingdom	49
Crew Agreements	50
Wages	51
Obligation of the shipowner in respect of seaworthiness	52
Crew Accommodation	52
Medical care and expenses	53
Repatriation and relief	53
4.3 South Africa	54
Crew Agreements	54
Minimum Age	55
Wages	55
Repatriation and relief	56
Medical Care	57
The Merchant Shipping Bill 2020	58
CHAPTER FIVE: CONCLUSION	
5.1 The Maritime Labour Convention	64
5.2 The United States of America	66
5.3 The United Kingdom	70
5.4 South Africa	72
5.5 Final remarks and observations	74
BIBLIOGRAPHY	76

#### **DECLARATION OF ORIGINALITY**

- I, Mikaela Jauden Joseph, declare that:
- The research reported in this dissertation, except where otherwise indicated, is my original work.
- ii. This dissertation has not been submitted for any degree or examination at any other university.
- iii. This dissertation does not contain other persons' data, pictures, graphs or other information, unless specifically acknowledged as being sourced from other persons.
- iv. This dissertation does not contain other persons' writing, unless specifically acknowledged as being sourced from other researchers. Where other sources have been quoted, then:
  - a. their words have been re-written but the general information attributed to them has been referenced;
  - b. where their exact words have been used, their writing has been placed inside quotation marks, and referenced.
- v. Where I have reproduced a publication of which I am author, co-author or editor, I have indicated in detail which part of the publication was actually written by myself alone and have fully referenced such publications.
- vi. This dissertation does not contain text, graphics or tables copied and pasted from the internet, unless specifically acknowledged, and the source being detailed in the dissertation and in the Reference sections.



Mikaela Jauden Joseph (215002928)

Signed at Durban on this 06 day of January 2022.

## **ACKNOWLEDGEMENTS**

I would like to acknowledge and give my warmest thanks to my supervisor Mrs Deepa Lamb who made this work possible. Her guidance, constant support and editorial assistance have carried me through all the stages of my dissertation.

To my parents, thank you for the continuous and unwavering support, love and understanding when undertaking my research and writing of my dissertation. Your prayer, continuous financial and emotional support and sacrifices are what have sustained me this far. Thank you for never giving up on me and never allowing me to give up on my dreams and aspirations.

I would like to give special thanks to Iman Mia and Kiara Philip, true friendship is difficult to find. Thank you for the constant support and motivation throughout this journey. Your moral support is what has helped me push past the many obstacles I have faced.

Finally, I would like to thank the Almighty, for getting me through all the difficulties I have faced. I have experienced your guidance day by day.

# **LIST OF ABBREVIATIONS**

AJRA Admiralty Jurisdiction Regulation Act 105 of 1983

CCMA Commission for Conciliation, Mediation and Arbitration (SA)

CMTP Comprehensive Maritime Transport Policy (SA)

FELA Federal Employers' Liability Act of 1908 (USA)

ILO International Labour Organisation

IMO International Maritime Organization

MARPOL International Convention for the Prevention of Pollution from

Ships, 73/78

MCA Maritime and Coastguard Agency (UK)

MLC Maritime Labour Convention

SA South Africa

SOLAS International Convention for Safety of Life at Sea, 1974

STCW International Convention on Standards of Training, Certification

and Watchkeeping, 1978

PSC Port State Control (USA)

UK United Kingdom

UNCLOS United Nations Convention on the Law of the Sea,1982

UNCTAD United Nations Conference on Trade and Development

USA United States of America

USCG United States Coast Guard

WW1 World War I

## **CHAPTER ONE: INTRODUCTION**

# 1.1 Introduction:

Maritime laws regulating waterborne transportation existed in some form since seafarers first set sail. Over time, these laws have evolved to adequately cater for vast technological advancements and changing global commercial practices. Many international treaties and conventions have been developed to help unify maritime laws. Given the development of maritime laws and the role of seafarers being integral to the operation of the global maritime trade industry, it would be expected that the rights of seafarers and the liability of owners would be clearly established. However, when analysing the extent of shipowner liability for the personal injuries sustained by seafarers there still appears to be some level of uncertainty as to the rights of seafarers and the extent of the shipowner's liability.<sup>1</sup>

During their employment, on board a vessel or while carrying out their respective duties, seafarers may be injured in several ways namely:<sup>2</sup>

- i. A seafarer may be injured in the service of the ship by accident, through no fault of the owners, master or crew.
- ii. A seafarer may be injured through the negligence of another member of the ship's company.
- iii. A seafarer may be injured through the breaking of the rigging, or of some appliance of the ship as a result of its defective condition.
- iv. A seafarer may have his health injured, either temporarily or permanently, through the lack of proper provisions and medicines on board the ship, or as a result of the master's failure to furnish him with the necessary provisions or medication on board.

<sup>&</sup>lt;sup>1</sup> Fitz- Henry Smith, Jr "Liability in the Admiralty for Injuries to Seamen." (1906) 19 (1) Harvard Law Review. P 418.

<sup>&</sup>lt;sup>2</sup> Fitz-Henry Smith, Jr "Liability in the Admiralty for injuries to Seamen." (1906) 19 (1) *Harvard Law Review.* P 418.

- v. An original injury to a seafarer may be aggravated or made permanent by the failure of the owners or the officers of the ship to properly care for or treat the injury.
- vi. A seafarer may be injured by a physical act of violence committed upon him by
  (a) the master, (b) one of the subordinate officers of the ship or, (c) another
  seafarer.

When a seafarer has been injured in one of the ways mentioned above, and it has been found that the shipowner is liable, the seafarer will be entitled to compensation for his maintenance and cure, as well as to his wages and his repatriation.3 This remedy available to seafarers is one of the earliest remedies available to a seafarer and can be found both in national laws as well as in the maritime codes. Despite the existence of this remedy, there is some uncertainty and inconsistency in the application of the remedy, namely, in relation to the duration of the right of the seafarer to be cured at the expense of the shipowner.<sup>5</sup> When looking at some of the earlier treaties and national legislation it appears that the word "cure" was used in the sense of "care" and not healing.<sup>6</sup> As the law developed it appeared that some jurisdictions would compensate the injured seafarer for a reasonable time after the termination of employment, while other jurisdictions would compensate the injured seafarer for the duration of the voyage and his repatriation.7 As a result of these differences in interpretation and the legislation being unclear as to exactly when the shipowners liability towards the seafarer ends, it appears that some seafarers are in a less favoured position depending on the domestic workmen's compensation legislation within their jurisdiction.<sup>8</sup> In view of this situation it would seem better policy to require the shipowner to maintain and care for a seafarer who has been injured in his service,

<sup>&</sup>lt;sup>3</sup>Fitz- Henry Smith, Jr "Liability in the Admiralty for Injuries to Seamen." (1906) 19 (1) *Harvard Law Review*. P 419.

<sup>&</sup>lt;sup>4</sup> The doctrine of maintenance and cure is rooted in *Article VI* of the *Rolls of Oleron* and have been developed over time to form some of the older maritime conventions such as, the *International Convention relating to the Limitation of Liability of Owners of Sea-Going Ships (Brussels 1957)* as well as the *1979 Protocol Amending the Convention, ILO Convention Concerning the Repatriation of Seafarers (1987).* 

<sup>&</sup>lt;sup>5</sup> "Admiralty. Extent of Liability to Injured Seaman." (1921) 30 (8) *The Yale Law Journal*. P860.

<sup>&</sup>lt;sup>6</sup> The earlier national legislation and treaties which were interpreted in this manner is the *Jones Act of 1920* as well as the *International Convention relating to the Limitation of Liability of Owners of Sea-Going Ships* (Brussels 1957).

<sup>&</sup>lt;sup>7</sup> "Admiralty. Extent of Liability to Injured Seaman." (1921) 30 (8) The Yale Law Journal. P860.

<sup>&</sup>lt;sup>8</sup> "Admiralty. Extent of Liability to Injured Seaman." (1921) 30 (8) The Yale Law Journal. P860.

not only for a reasonable time after the termination of employment, but until the seafarer is able to return to work or it has been deemed that he is unable to work.

In 2006 the International Labour Organisation (ILO) adopted the Maritime Labour Convention (MLC). The purpose of the MLC is to establish international standards and rules applicable to seafarers to ensure that they are treated in a fair and uniform manner. The MLC has outlined minimum requirements for the entry of seafarers to ships, employment conditions, conditions in relation to crew, welfare facilities, food and logistics, community services, health care and health protection. The MLC was implemented on August 20, 2013 and one year later had the registration of 30 countries. By November 2016, 80 countries that accounted for 87% of the international shipping community had ratified the Convention.

The Convention contains 'soft law' which is a set of obligations that were created by the negotiating parties and are not legally binding. The 'hard law' contained in the Convention is a set of legally binding rules and regulations.<sup>11</sup> Through these laws the MLC aims to establish a level playing field for countries and shipowners, ensuring they are committed to providing decent working and living conditions for seafarers, protecting them from unfair competition on the part of substandard ships.<sup>12</sup>

When looking at the MLC it is important to note that the Convention itself cannot be cited as authoritative legislation because the precise requirements of the convention are contained in the national laws or regulations adopted by each country to implement the MLC.<sup>13</sup> Therefore while the MLC appears to be a promising international convention aimed at ensuring the fair treatment of seafarers, its effectiveness in protecting seafarers can only be established by looking at how it has been applied through the national laws of countries.

<sup>&</sup>lt;sup>9</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble

<sup>&</sup>lt;sup>10</sup> https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:80001:0 accessed on 13 August 2022.

<sup>&</sup>lt;sup>11</sup> A Mokhtari, M Lafteh, R Hematjoo "Investigation of the Maritime Labor Convention and its legal effects for countries" (2017) *Research Gate*. P 152.

<sup>&</sup>lt;sup>12</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>&</sup>lt;sup>13</sup> "Maritime Labour Convention, 2006 Frequently Asked Questions" (2012) *International Labour Organisation*. P 1.

## 1.2 Scope and limitations of the study

The central objective of this dissertation is to establish whether the MLC has been successful in ensuring the fair and equal treatment of seafarers in relation to shipowner liability for personal injuries sustained by seafarers while in their employ. To determine whether the MLC has been successful in protecting the rights of seafarers, this dissertation will examine the national legislation in three jurisdictions, namely, South Africa (SA), the United Kingdom (UK) and the United States of America (USA). This dissertation will examine the national legislation pertaining to the extent of shipowner liability of injured seafarers in the abovementioned jurisdictions and the extent of shipowner liability in national legislation after the implementation of the MLC in the abovementioned jurisdictions. The legislation and international conventions that will be examined in this dissertation are:

- a) The Maritime Labour Convention<sup>14</sup>,
- b) UNCLOS<sup>15</sup>,
- c) the Merchant Shipping Act<sup>16</sup>,
- d) the Merchant Shipping Amendment Act<sup>17</sup>,
- e) the Merchant Shipping Act<sup>18</sup>,
- f) the Merchant Shipping Act (liability of shipowners and others)<sup>19</sup>,
- g) the Merchant Shipping and Maritime Security Act<sup>20</sup>,
- h) the Jones Act<sup>21</sup>,
- i) the Merchant Marine Act<sup>22</sup>

<sup>&</sup>lt;sup>14</sup> The Maritime Labour Convention, 2006.

 $<sup>^{\</sup>rm 15}$  United Nations Convention on the Law of the Sea , 1982.

<sup>&</sup>lt;sup>16</sup> SA Merchant Shipping Act 57 of 1951.

<sup>&</sup>lt;sup>17</sup> SA Merchant Shipping Amendment Act 12 of 2015.

<sup>&</sup>lt;sup>18</sup> UK Merchant Shipping Act Commencement no 1 of 1970.

<sup>&</sup>lt;sup>19</sup> UK Merchant Shipping (liability of shipowners and others) Act 1958.

<sup>&</sup>lt;sup>20</sup> UK Merchant Shipping and Maritime Security Act Of 1997.

<sup>&</sup>lt;sup>21</sup> US Jones Act of 1920.

<sup>&</sup>lt;sup>22</sup> US Merchant Marine Act Of 1970.

All other international conventions and national legislation will fall outside the scope of this dissertation.

For the purposes of this dissertation; shipowner liability will be defined as instances in which the shipowner has a responsibility or duty to compensate seafarers injured on board a vessel while in the course and scope of their duties. As mentioned earlier in this chapter, a seafarer may be injured in several ways leading to a shipowner being liable for compensation. An examination of the types of injuries resulting in shipowner liability falls outside the scope of this dissertation. It will focus instead on an examination of the extent and duration of shipowner liability in instances where it has already been established that the shipowner is liable for the personal injuries sustained by seafarers in their employ and have accepted liability for injury. Liability refers to the duty that the shipowner has to compensate a seafarer who has been injured, and it will be presumed that the shipowner has accepted liability for the injury.

For the purposes of this dissertation; injury will refer to the physical harm, including sickness sustained by a seafarer. This will also include instances where a seafarer's health has been injured due to a lack of care or due to improper care (where a prior physical injury or sickness has worsened due to improper care or no care). Psychological harm sustained by a seafarer falls outside the scope of this dissertation.

For the purposes of this dissertation; compensation will refer to the act of providing a seafarer with money or wages, including payment of costs such as medical and disability expenses that arise as a result of their injuries.

Damages for the purposes of this dissertation, will include monetary loss sustained by seafarers, such as loss of wages as a result of being unable to work or to continue working, and funeral costs in the instance of death.

# 1.3 Research Objectives and Rationale

The main objective of this dissertation is to establish whether the MLC has been successful in ensuring the fair and equal treatment of seafarers in relation to shipowner liability. This dissertation examines national legislation pertaining to the extent of shipowner liability of injured seafarers both prior to, and after the implementation of the MLC. This dissertation aims to establish whether the MLC has been successful in adequately protecting the rights of injured seafarers or whether seafarers in certain

jurisdictions are still being treated unfairly. This dissertation will look at three jurisdictions, namely South Africa, the United Kingdom (UK) and the United States of America (USA). Given South Africa's close relationship with English law, the history and development of South African admiralty law, pertaining to the protection of seafarers, cannot be examined in isolation from English law. Furthermore, this dissertation will examine the national law in the United States of America (USA) because the USA is one of the oldest shipping nations and has played an integral role in the development of admiralty laws pertaining to the protection of seafarers. This dissertation will examine the history of national legislation pertaining to the protection of seafarers in the respective three jurisdictions and the current national legislation pertaining to the protection of seafarers in the respective jurisdictions after the implementation of the MLC. This will determine if the MLC has been successful in bridging the gap that existed and whether it has provided clarity as to when the liability of a shipowner comes to an end.

In order to achieve the objectives of this dissertation several international treaties and conventions as well as domestic legislation listed above under the scope of this study will be examined. The Maritime Labour Convention<sup>23</sup> as well as UNCLOS<sup>24</sup> will be examined because these international conventions have been instrumental in the protection of seafarers' rights and have been ratified by most, if not all shipping nations. In addition, these conventions serve as the basis on which domestic legislation is developed in countries which have ratified the conventions.

The domestic legislation of the abovementioned jurisdictions will also be examined. In South Africa the Merchant Shipping Act<sup>25</sup> will be the first set of domestic legislation utilised to determine the extent of which the act protected seafarers. This Act will be examined because it was implemented well before the MLC was developed and therefore provides a clear indication of the gap that existed in the protection of seafarers' rights prior to the implementation of the MLC. Thereafter the Merchant Shipping Amendment Act<sup>26</sup> will be examined because it was implemented after the MLC and gives effect to it in the Act. These two pieces of domestic South African

<sup>&</sup>lt;sup>23</sup> The Maritime Labour Convention, 2006.

<sup>&</sup>lt;sup>24</sup> United Nations Convention on the Sea Law, 1982.

<sup>&</sup>lt;sup>25</sup> Act 57 of 1951.

<sup>&</sup>lt;sup>26</sup> Act 12 of 2015

legislation will be examined to ascertain whether the MLC has succeeded in bridging the gap between seafarers' rights and whether these are adequately protected within this jurisdiction.

The United Kingdom (UK) Merchant Shipping Act<sup>27</sup> and the Merchant Shipping Act (liability of shipowners and others)<sup>28</sup>, will be examined in order to establish the shortfalls, gaps and extent of the protection of seafarers' rights prior to the implementation of UNCLOS and the MLC. Thereafter, the more recent UK Merchant Shipping and Maritime Security Act<sup>29</sup> pertaining to the protection of seafarers, will be examined. Such examination will ascertain whether previous gaps in the law have been bridged or whether seafarers' rights in the UK are still inadequately protected despite the coming into effect of the MLC.

The United States of America (USA) Jones Act<sup>30</sup> and the Merchant Marine Act<sup>31</sup>, will be examined to establish the extent to which seafarers' rights were protected prior to the implementation of the MLC. Thereafter, the current laws pertaining to the protection of seafarers', will be examined to establish whether seafarers' rights in the United States are adequately protected and whether the MLC has been successfully implemented within this jurisdiction.

This dissertation will focus on the extent and duration of shipowner liability and will examine the application and implementation of such liability within the three jurisdictions identified for the purposes of this study.

#### 1.3.1 Presentation of key research questions

This research aims to answer the following key research questions:

#### 1.3.1.1 Key research question 1

To what extent were seafarers' rights protected when they were injured during their employment and at what point did a shipowner's liability end when a seafarer was injured before coming into operation of the MLC?

<sup>&</sup>lt;sup>27</sup> Commencement no 1 of 1970

<sup>&</sup>lt;sup>28</sup> Act 1958.

<sup>&</sup>lt;sup>29</sup> 1997.

<sup>&</sup>lt;sup>30</sup> 1920.

<sup>&</sup>lt;sup>31</sup> of 1970.

## Sub- research questions

To what extent were seafarers' rights protected when they were injured during their employment under South African legislation before the implementation of the MLC?

At what point did a shipowner's liability end under South African law before the MLC was implemented?

To what extent were seafarers' rights protected when they were injured during their employment under the United Kingdom's legislation before the implementation of the MLC?

At what point did a shipowner's liability end under the legislation in the United Kingdom before the MLC was implemented?

To what extent were seafarers' rights protected when they were injured during their employment under USA legislation before the implementation of the MLC?

At what point did a shipowner's liability end under USA law before the MLC was implemented?

## 1.3.1.2 Key research question 2

To what extent are seafarers' rights currently protected when they are injured during their employment and at what point does a shipowner's liability end once a seafarer has been injured after the implementation of the MLC?

#### Sub research questions

To what extent are seafarers' rights currently protected under South African law when they are injured during their employment?

At what point does a shipowner's liability end once a seafarer has been injured under South African law after the implementation of the MLC?

To what extent are seafarers' rights currently protected under English law when they are injured during their employment?

At what point does a shipowner's liability end once a seafarer has been injured under English law after the implementation of the MLC?

To what extent are seafarers' rights currently protected under USA law when they are injured during their employment?

At what point does a shipowner's liability end once a seafarer has been injured under USA law after the implementation of the MLC?

# 1.3.1.3 Key research question 3

Has the implementation of the MLC been successful in ensuring the protection of seafarer's rights when they are injured, and has the MLC clearly identified the end of a shipowner's liability?

#### Sub research questions

Has the implementation of the MLC been successful in ensuring the protection of seafarer's rights when they are injured and has the duration of a shipowner's liability been clearly defined under South African law?

Has the implementation of the MLC been successful in ensuring the protection of seafarer's rights when they are injured and has the duration of a shipowner's liability been clearly defined under English law?

Has the implementation of the MLC been successful in ensuring the protection of seafarer's rights when they are injured and has the duration of a shipowner's liability been clearly defined under American law?

#### 1.3.1.4 Key research question 4

Has the MLC been successful in ensuring seafarers are treated fairly and are at an equal playing field and has it created uniformity with regard to their treatment?

#### Sub research question

After looking at South African law, English law and American law after the implementation of the MLC are seafarers from all three jurisdictions at an equal playing field within the three jurisdictions identified for the purposes of this study?

Does an examination of South African, English and USA laws giving effect to the MLC, indicate fair treatment of seafarers in all three jurisdictions?

# 1.4 Research methodology and limitations

The research methodology used in this dissertation is predominantly desktop-based and makes use of comparative research methodology. The researcher finds that this method is best suited to this dissertation because it involves the collation of existing national legislation and international conventions as well as journal articles and periodicals. In addition, comparative research methodology is used as the three identified jurisdictions are compared to each other and compared in relation to the Maritime Labour Convention. The researcher has also utilised primary and secondary sources of research. The primary sources of research include the Maritime Labour Convention<sup>32</sup>, UNCLOS<sup>33</sup>, the Merchant Shipping Amendment<sup>34</sup> (South Africa), the Merchant Shipping Act<sup>35</sup>(South Africa), the Merchant Shipping Act<sup>36</sup> (United Kingdom), the Merchant Shipping and Maritime Security Act<sup>37</sup> (United Kingdom), the Merchant Shipping Act<sup>38</sup> (United Kingdom), the Jones Act<sup>39</sup> (United States) and the Merchant Marine Act<sup>40</sup> (United States). The researcher also made use of information from secondary sources which include textbooks and online literature, e- journals, websites, theses and online newspapers.

The researcher has encountered some limitations in the course of this research which may, possibly have affected its outcome. The most challenging of these outcomes is the scarcity of the older national statutes which were utilised before amendments and the implementation of the MLC. For this reason, the researcher has had to rely on journal articles for further explanation on some of the older national legislation.

#### 1.5 Structure of the dissertation

This dissertation is broken down into five chapters. Chapter one introduces the problem topic and provides the background to the problem. It states the research objectives, the key research questions, the research methodology and the limitations.

<sup>&</sup>lt;sup>32</sup> The Maritime Labour Convention, 2006.

<sup>&</sup>lt;sup>33</sup> United Nations Convention on the Sea Law, 1982.

<sup>&</sup>lt;sup>34</sup> Act 12 of 2015.

<sup>&</sup>lt;sup>35</sup> Act 57 of 1951.

<sup>&</sup>lt;sup>36</sup> Commencement no 1 of 1970

<sup>&</sup>lt;sup>37</sup> 1997

<sup>&</sup>lt;sup>38</sup> (Liability of shipowners and others) Act 1958.

<sup>&</sup>lt;sup>39</sup> Merchant Marine Act of 1920.

<sup>&</sup>lt;sup>40</sup> Of 1970.

Chapter two examines the MLC, it provides information on the aims of the MLC, how it was developed, the implementation of the MLC and the provisions for shipowner liability in respect of injured seafarers.

Chapter 3 examines the history of the national laws in the respective three jurisdictions before the implementation of the MLC and analyses the extent to which an injured seafarers' rights were protected and where a shipowner's liability ends.

Chapter 4 examines the national laws in the respective three jurisdictions after the implementation of the MLC and analyses the extent to which an injured seafarers' rights are now protected and examines exactly where a shipowner's liability ends.

Chapter 5 concludes the dissertation. It restates the key research questions and the resulting conclusions.

#### **CHAPTER TWO: THE MARITIME LABOUR CONVENTION**

## 2.1 Introduction

The International Maritime Organisation (IMO): International Maritime Law Institute has noted the importance of international maritime labour law.<sup>41</sup> One of the main difficulties of international maritime labour law is achieving a balance between shipowners and seafarers' interests.<sup>42</sup> The International Labour Organisation (ILO) has taken on the task of attempting to balance the interests of both parties, an initiative that has engaged the ILO for decades.<sup>43</sup>

During the period 1920 to 1996, the ILO adopted sixty-six Conventions related to seafarers' working conditions; over the years, the Conventions have been criticised because they have become less relevant.<sup>44</sup> The Conventions were critiqued because they were enforced irregularly and often not adhered to; thus, the ILO responded by reviewing the maritime labour Conventions to make them relevant to the industry's current needs.<sup>45</sup> As a result, the ILO adopted the Maritime Labour Convention, 2006 (MLC), an international convention to consolidate the pre-existing legal instruments on maritime labour law.<sup>46</sup>

The MLC is an international labour convention adopted by the ILO in February 2006 during an International Labour Conference held in Geneva.<sup>47</sup> The MLC encompasses seafarers' rights and ensures that they have satisfactory working conditions; in addition to this, the MLC promotes fair competition between shipowners.<sup>48</sup> The Convention applies to all "seafarers",<sup>49</sup> which the Convention defines as any person

<sup>&</sup>lt;sup>41</sup> Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO. Appendix 2.

<sup>&</sup>lt;sup>42</sup> Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO. Appendix 2

<sup>&</sup>lt;sup>43</sup>Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO. Appendix 2

<sup>&</sup>lt;sup>44</sup> Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO. Appendix 2.

<sup>&</sup>lt;sup>45</sup> Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO. Appendix 2

<sup>&</sup>lt;sup>46</sup> Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO. Appendix 2

<sup>&</sup>lt;sup>47</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

 $<sup>^{\</sup>rm 48} Maritime$  Labour Convention, 2006 as amended, (2016) Title 1

<sup>&</sup>lt;sup>49</sup> Maritime Labour Convention, 2006 as amended, (2016) Article II, Paragraph 1 (f).

employed or engaged in work in any capacity on board a vessel.<sup>50</sup> The Convention was developed to be used as an international legal instrument complementing preceding international conventions such as the International Convention for Safety of Life at Sea, 1974 as amended (SOLAS), the International Convention on Standards of Training, Certification and Watchkeeping, 1978 as amended (STCW) and the International Convention for the Prevention of Pollution from Ships, 73/78 (MARPOL).<sup>51</sup>

The MLC, 2006 is a comprehensive legal instrument that contains a complete set of international standards contained in preceding labour instruments.<sup>52</sup> The MLC has a modern format that aligns with current conditions and language; it consolidates all but four of the existing maritime labour instruments into a concise convention.<sup>53</sup>

A new convention was required because many flag states do not exercise adequate jurisdiction or control over the ships that fly their flag, as required by international law.<sup>54</sup> As a result, seafarers working onboard these vessels are subjected to unacceptable working conditions that are hazardous to their health, well-being and safety, and the safety of the ship itself.<sup>55</sup> Due to the maritime industry's international nature, many seafarers spend much of their time working outside of their resident countries; in addition to this, their employers are frequently not based in their resident countries.<sup>56</sup> As a result, there is a need for practical international standards, particularly for seafarers' protection, in the maritime industry.<sup>57</sup> However, it is not enough to have these standards solely implemented globally; they should also be implemented nationally by flag states.<sup>58</sup> While it is of utmost importance to protect seafarers' rights, it is vital to recognise flag states and shipowners who ensure that they safeguard

-

<sup>&</sup>lt;sup>50</sup> MLC 2006, Article II, Paragraph 2.

<sup>&</sup>lt;sup>51</sup> Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO. Appendix 2.

<sup>&</sup>lt;sup>52</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>&</sup>lt;sup>53</sup> The MLC, 2006 does not revise the Convention addressing seafarers' identity documents of 2003 (Convention No. 185) and the 1958 Convention that it revises (Convention No. 108) are not consolidated in the New Convention; nor are the Seafarers' Pension Convention, 1946 (No. 71) and the Minimum Age (Trimmers and stokers) Convention, 1921 (No. 15).

<sup>&</sup>lt;sup>54</sup>United Nations Convention on the Law of the Sea, 1982 (UNCLOS), Article 94

<sup>&</sup>lt;sup>55</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 25 May 2020).

<sup>&</sup>lt;sup>56</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions page 4-5.

<sup>&</sup>lt;sup>57</sup>International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions page 4-5.

<sup>&</sup>lt;sup>58</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 25 May 2020).

seafarers and ensure decent working conditions.<sup>59</sup> These shipowners and flag states face unfair competition, as they are unfairly disadvantaged by being undercut by shipowners who operate substandard ships.<sup>60</sup>

As a result of a joint resolution by international seafarers' and shipowner organisations in 2001, the ILO decided to proceed with the MLC.<sup>61</sup> It was noted that the shipping industry is the world's first genuinely global industry and, therefore, the industry, required an appropriate international regulatory response based on applicable global standards.<sup>62</sup> The multitude of existing maritime conventions made it difficult for governments to ratify and enforce all these conventions, resulting in an ILO initiative to develop an instrument that would consolidate existing maritime instruments.<sup>63</sup> The other problems experienced with the existing maritime conventions were that many were outdated and did not adequately address the current working and living conditions onboard vessels.<sup>64</sup> Additionally, many of the existing conventions had low levels of ratification.<sup>65</sup> There was also a need to develop effective enforcement and compliance systems that would prevent the use of substandard vessels and efficiently enforce the standards for ship safety and security adopted within the International Maritime Organisation framework (IMO).<sup>66</sup> The MLC was thus designed to address these concerns.<sup>67</sup>

## 2.2 The primary aims of the Maritime Labour Convention

There are two basic aims of the MLC, namely:68

I. To ensure comprehensive worldwide protection of the rights of seafarers.

<sup>&</sup>lt;sup>59</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions page 4-5.

<sup>60</sup> Labour standards: Basic facts on the Maritime Labour Convention 2006. 

<a href="https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS">https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS</a> 219665/lang-en/index.htm. (Accessed 22 September 2020).

<sup>&</sup>lt;sup>61</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>&</sup>lt;sup>62</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>&</sup>lt;sup>63</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>&</sup>lt;sup>64</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>65</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>66</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>&</sup>lt;sup>67</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

<sup>&</sup>lt;sup>68</sup> Maritime Labour Convention, 2006 as amended, (2016) Preamble.

II. To establish a level playing field for countries and shipowners committed to providing decent working and living conditions for seafarers, protecting them from the unfair competition on the part of substandard ships.

The MLC is a comprehensive convention that sets out seafarers' rights and ensures they have decent working conditions on almost every aspect of their working and living conditions.<sup>69</sup> The MLC addresses issues such as minimum age of employment, employment agreements, the hours of work and hours of rest, payment of wages, paid annual leave, repatriation at the end of the contract, medical care onboard vessels, accommodation, food and catering, health and safety protections, and accident prevention as well as seafarers' complaint handling.<sup>70</sup>

#### 2.3 The basic structure of the MLC

The MLC contains three parts that are related, namely the Articles, the Code and the Regulations.<sup>71</sup> The Articles and the Regulations set out the basic principles of the Convention and Member States' obligations.<sup>72</sup> The Code comprises Part A and Part B, containing information on how the Member States implement the Regulations.<sup>73</sup> The Regulations and the Code include basic standards and guidelines of implementation and are organised into five Titles.<sup>74</sup> Title 1, "Minimum requirements for seafarers", encompasses requirements related to minimum age, medical certificates, training and recruitment.<sup>75</sup> Title 2, "Conditions of employment," contains guidelines on seafarers employment contracts, leave entitlement, wages, repatriation, hours of work, rest, and career and skills development.<sup>76</sup> Title 3, "Accommodation, recreational facilities, food and catering," deals with the recreational facilities required onboard, the

<sup>69</sup> Labour standards: Basic facts on the Maritime Labour Convention 2006. https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS 219665/lang-en/index.htm. (Accessed 22 September 2020).

 $<sup>^{70}</sup>$  Maritime Labour Convention, 2006 as amended, (2016) see Regulations 1.1, 2.5, 31 and 3.2

<sup>&</sup>lt;sup>71</sup> Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>72</sup> Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>73</sup> Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>74</sup> Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>75</sup> Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>76</sup>Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

basic standards of accommodation and food and catering requirements.<sup>77</sup> Title 4, "Health protection, medical care, welfare and social security protection," this title comprises provisions relating to medical care, health and safety protection, shipowner liability, accident prevention and shore-based welfare facilities.<sup>78</sup> Lastly, Title 5, "Compliance and enforcement," pertains to flag state responsibilities such as maritime labour certificates and compliance, labour supply responsibilities, and Port State responsibilities.<sup>79</sup>

# 2.4 How the MLC aims to protect more seafarers

The MLC has been designed to achieve a higher level of ratification than the conventions preceding it.<sup>80</sup> The MLC also aims to protect seafarers' rights who work on vessels registered in states that have not ratified the MLC.<sup>81</sup> Additionally, the MLC extends the scope of its protection to the rights of all persons working onboard a vessel.<sup>82</sup> Before the coming into operation of the MLC, it was not clear whether people working onboard a ship but not directly involved in the navigation or operation of the vessel would be considered seafarers for preceding conventions.<sup>83</sup>

The MLC aims to ensure that there is "continuous compliance" with its measures at both an international and national level.<sup>84</sup> This process of continuous compliance is initiated by ensuring that seafarers are correctly informed of their rights and any remedies available to them in instances of non-compliance with the provisions of the MLC.<sup>85</sup> In addition to this, shipowners that operate vessels engaged in international voyages are required to develop and implement plans ensuring that national laws and

<sup>&</sup>lt;sup>77</sup>Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>78</sup>Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>79</sup> Maritime Labour Convention, 2006 as amended, (2016) see Explanatory Note to the Regulations and Code of the Maritime Labour Convention.

<sup>&</sup>lt;sup>80</sup> Labour standards: Basic facts on the Maritime Labour Convention 2006. https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS\_219665/lang-en/index.htm. (Accessed 22 September 2020).

<sup>&</sup>lt;sup>81</sup> ILO: Adoption of an instrument to consolidate maritime labour standards, Report I (1A), at 42.

<sup>&</sup>lt;sup>82</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 301.

<sup>&</sup>lt;sup>83</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 25 May 2020).

 $<sup>^{84}</sup>$  Maritime Labour Convention, 2006 as amended, (2016) see Title 5.

<sup>85</sup> Maritime Labour Convention, 2006 as amended, (2016) see Title 5.

regulations implementing provisions of the MLC are being complied with.<sup>86</sup> To ensure that this process is carried out effectively and that shipowners' plans are being implemented, the master of the vessel is mandated to keep a proper record of implementing these plans as evidence of adherence to the MLC.<sup>87</sup> To ensure 'continuous compliance', responsibility has been placed on the Flag State to ensure that shipowners meet national and international compliance.<sup>88</sup> Consequently, the Flag State is responsible for reviewing the shipowners' plans and verifying that the master has correctly implemented the plans.<sup>89</sup>

Vessels are further required to carry a Maritime Labour Certificate and a Declaration of Maritime Labour Compliance on board.<sup>90</sup> Flag states will be required to carry out regular assessments to determine the quality and efficacy of their national systems of compliance.<sup>91</sup> Additionally, the flag state's reports sent to the ILO must contain information on the inspections carried out and methods utilised to do this.<sup>92</sup>

## 2.5 Minimum standards for seafarers

## 2.5.1 Minimum Age

The MLC prohibits the employment and work of a person under the age of 16 onboard a vessel <sup>93</sup>; furthermore, it prohibits anyone under 18 from working a night shift <sup>94</sup>. The MLC's age requirements are similar to those contained in the Seafarers' Hours of Work and the Manning of Ships Convention 1996 (no. 180). <sup>95</sup> In addition, the MLC excludes work that may endanger the health or safety of a seafarer under the age of 18; the national laws or competent authority will determine the type of work prohibited. <sup>96</sup>

<sup>&</sup>lt;sup>86</sup> Maritime Labour Convention, 2006 as amended, (2016) see Title 5.

<sup>&</sup>lt;sup>87</sup> Maritime Labour Convention, 2006 as amended, (2016) see Title 5

<sup>88</sup> Maritime Labour Convention, 2006 as amended, (2016) see Title 5

<sup>&</sup>lt;sup>89</sup> Maritime Labour Convention, 2006 as amended, (2016) see Title 5

<sup>&</sup>lt;sup>90</sup> Maritime Labour Convention, 2006 as amended, (2016) see Title 5

<sup>&</sup>lt;sup>91</sup>Maritime Labour Convention, 2006 as amended, (2016) see Title 5

<sup>&</sup>lt;sup>92</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 25 May 2020).

<sup>&</sup>lt;sup>93</sup> MLC, 2006, Standard A1.1, paragraph 1.

<sup>&</sup>lt;sup>94</sup> MLC, 2006, Standard A1.1, paragraph 2.

<sup>&</sup>lt;sup>95</sup> Convention no 180, Articles 6 and 12.

<sup>&</sup>lt;sup>96</sup> MLC, 2006, Standard A1.1, paragraph 4.

Concerning the minimum age requirements, the MLC has not proposed any substantive changes to the pre-existing Conventions.<sup>97</sup>

# 2.5.2 Seafarers' employment agreements

Under the MLC, each member state must adopt national laws or regulations that ensure that vessels that fly its flag must comply with basic requirements regarding employment agreements. Standard A2.1 requires: (a) seafarers' must have an employment agreement signed by both the seafarer and shipowner or their representative, (b) seafarers must be allowed to examine and, where necessary, seek advice on the agreement before signing it, (c) both parties to the contract must have a signed copy of such, (d) conditions of employment must be easily accessible to seafarers while onboard and the employment contract must be available to review by officers of a competent authority and (e) seafarers must be given a document recording their employment onboard the vessel. 99

The MLC has introduced several significant changes to the Seamen's Articles of Agreement Convention, 1926 (No. 22), namely: (a) the MLC explicitly requires member states to ensure seafarers have an employment agreement<sup>100</sup> (b) it requires that the conditions of employment must be available for review<sup>101</sup>, (c) it added health and social security protection benefits and entitlement to repatriation to the particulars contained in the employment agreement<sup>102</sup> and (d) the minimum notice for early termination of the contract shall not be earlier than seven days.<sup>103</sup>

#### 2.5.3 Wages

The MLC requires that wages are paid to seafarers in no greater than monthly intervals.<sup>104</sup> In addition, the MLC makes it mandatory for shipowners to ensure that

<sup>&</sup>lt;sup>97</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 303.

<sup>&</sup>lt;sup>98</sup> MLC, 2006, Standard A2.1.

<sup>&</sup>lt;sup>99</sup> MLC, 2006, Standard A2.1 paragraph 1.

<sup>&</sup>lt;sup>100</sup> MLC, 2006, Standard A2.1 paragraph 1(d).

<sup>&</sup>lt;sup>101</sup> MLC, 2006, Standard A2.1 paragraph 1 (e).

<sup>&</sup>lt;sup>102</sup> MLC, 2006, Standard A2.1 paragraphs 4(h), 4(i) and 4(j).

<sup>&</sup>lt;sup>103</sup> MLC, 2006, Standard A2.1 paragraph 5.

<sup>&</sup>lt;sup>104</sup> MLC, 2006, Standard A2.2 paragraph 1.

seafarers can transmit their wages to their families.<sup>105</sup> The MLC also contains guidelines as to the calculation and payment of wages<sup>106</sup> and minimum wages.<sup>107</sup>

Before implementing the MLC, the period in which wages were to be paid was contained in the Seafarers' Wages, Hours of Work and the Manning of Ships Recommendation, 1996 (No. 187).<sup>108</sup> The Seamen's Welfare in Ports Recommendation states that seafarers must have the ability to transfer their wages.<sup>109</sup> Including these requirements in the MLC has had a substantial change in the standards concerning seafarers' wages because the previous Conventions merely recommended these measures and did not make them mandatory.<sup>110</sup>

#### 2.5.4 Repatriation

The MLC ensures that member states repatriate seafarers in instances where (a) the employment agreement expires while the seafarer is abroad, (b) the employment agreement is terminated by the shipowner or the seafarer for a justified reason and (c) the seafarer is unable to or cannot be expected to carry out their duties as outlined in the employment agreement.<sup>111</sup>

The MLC has introduced three significant changes to the Repatriation of Seafarers Convention (Revised) 1987 (No. 166), which preceded it.<sup>112</sup> The MLC makes it compulsory for member states to ensure that vessels flying their flag provide financial security to guarantee seafarers are repatriated per the MLC.<sup>113</sup> The preceding Convention allowed for repatriation upon expiry of the notice given period, regardless of the reason for termination.<sup>114</sup> Conversely, the MLC provides repatriation in similar cases only when the seafarer has terminated the contract for a justified reason.<sup>115</sup>

<sup>&</sup>lt;sup>105</sup> MLC, 2006, Standard A2.2 paragraphs 3 and 4.

<sup>&</sup>lt;sup>106</sup> MLC, 2006, Guideline B2.2.2.

<sup>&</sup>lt;sup>107</sup> MLC, 2006, Guideline 2.2.3.

<sup>&</sup>lt;sup>108</sup> Recommendation No. 187, section 6(d).

<sup>&</sup>lt;sup>109</sup> Recommendation No. 48, section 10 (b).

<sup>&</sup>lt;sup>110</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 308.

<sup>&</sup>lt;sup>111</sup> MLC, 2006, Standard A.2.5, paragraph 1.

<sup>&</sup>lt;sup>112</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 310.

<sup>&</sup>lt;sup>113</sup> MLC, 2006, Regulation 2.5, paragraph 1.

<sup>&</sup>lt;sup>114</sup> Convention No.166, Article 2, paragraph 1(b).

<sup>&</sup>lt;sup>115</sup> MLC, 2006, Standard A2.5, paragraph 1(b)(ii).

Lastly, with regard to the MLC, the details of repatriation have been placed in non-mandatory Part B of the Code to guide implementation at a national level. 116

## 2.5.5 Compensation for the vessel's loss or foundering

Under the MLC, each member state must ensure that in instances of loss or foundering, the shipowner must compensate each seafarer for unemployment due to the vessel's loss or foundering.<sup>117</sup> In the preceding Conventions, the Unemployment Indemnity (Shipwreck) Convention, 1920 (No. 8), the amount of compensation was stipulated under Article 2, paragraph 2; however, in the MLC, the relating provision is contained in Guideline B.2.6.1 paragraph 1; therefore, it is no longer mandatory.<sup>118</sup>

#### 2.5.6 Medical care on board and ashore

Under the MLC, member states must ensure that vessels that fly their flag provide adequate protection to seafarers health and ensure seafarers have access to acceptable health care on board the vessel. In principle, the health care on board the vessel should be provided at no cost to the seafarer. As a minimum requirement, the national laws implemented by member states must provide (a) that every vessel carry a medical chest, medical equipment and medical guide, subject to regular inspection by a competent authority, (b) that ships carrying 100 or more people and engaged in international voyages of more than three days have a medical doctor on board and (c) ships that do not carry a medical doctor must have at least one seafarer on board who is in charge of medical care and administering medicine.

The Health Protection and Medical Care (Seafarers) Convention, 1987 (No.164), made it compulsory for shipowners to provide medical care and health protection to seafarers onboard the vessel free of charge. On the contrary, the MLC has made this obligation "more flexible" by adding the words "in principle". As a result, unlike

<sup>&</sup>lt;sup>116</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 310.

<sup>&</sup>lt;sup>117</sup> MLC, 2006, Standard A2.6, paragraph 1.

<sup>&</sup>lt;sup>118</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 311.

<sup>&</sup>lt;sup>119</sup> MLC, 2006, Regulation 4.1, paragraph 1.

<sup>&</sup>lt;sup>120</sup> MLC, 2006, Regulation 4.1, paragraph 2.

<sup>&</sup>lt;sup>121</sup> MLC, 2006, Standard A4.1, paragraph 4.

<sup>&</sup>lt;sup>122</sup> Convention No. 164, paragraph 4(d).

<sup>&</sup>lt;sup>123</sup> ILO: Adoption of an instrument to consolidate maritime labour standards, Report I (1A), at 42.

the preceding Convention, the MLC does not prevent shipowners from providing paid medical care and medication to seafarers. 124

# 2.5.7 Shipowner Liability

Under the MLC, a member state must ensure that shipowners provide the following minimum standards concerning seafarer health protection and medical care, (a) shipowners are liable to bear the costs for injury and sickness of seafarers working onboard their vessel from the date of commencing duty till they are deemed duly repatriated, (b) shipowners must provide financial security to ensure seafarers are duly compensated in the event of death or long-term disability due to an occupational injury, illness or hazard, (c) the shipowner is liable for medical expenses, and board and lodging while the seafarer is away from home until the seafarer has recovered or the injury has been declared permanent, (d) in the instance of death of a seafarer on board or ashore the shipowner will be liable for the burial expenses for the period of engagement.<sup>125</sup>

Compared to the preceding Convention, the Shipowners' Liability (Sick and Injured Seamen) Convention, 1936 (No.55), the MLC prescribes an additional liability onto shipowners. 126 The MLC requires shipowners to provide financial security to ensure that seafarers are compensated in instances of death, long-term disability, occupational injury, illness or hazard. 127

# 2.6 How the MLC aims to protect shipowners

The provisions in the 2006 MLC aim to protect seafarers' rights and protect the interests of complying shipowners. 128 The MLC seeks to safeguard these shipowners' interests by implementing what is known as the "no more favourable treatment clause"

<sup>&</sup>lt;sup>124</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 316.

<sup>&</sup>lt;sup>125</sup> MLC, 2006, Standard A4.2, paragraph 1.

<sup>&</sup>lt;sup>126</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 317.

<sup>&</sup>lt;sup>127</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 317.

<sup>&</sup>lt;sup>128</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 25 May 2020).

that can be found in Article V, paragraph 7<sup>129</sup> of the MLC.<sup>130</sup> The objective of the "no more favourable treatment clause" is to ensure a "level playing field" among shipowners so that ships that operate in countries that have ratified the Convention will not be at a competitive disadvantage to vessels that operate in countries that have not ratified the Convention.<sup>131</sup>

At first glance, Article V, paragraph 7, gives the impression that it could be applicable in several scenarios.<sup>132</sup> In practice, however, it is implemented through port State control under regulation 5.2.1<sup>133</sup>, in respect of ships that fly a foreign flag and call at a port of a country that has ratified the Convention.<sup>134</sup> A closer examination of regulation 5.2.1<sup>135</sup> reveals that the "no more favourable clause" plays an instrumental role in protecting seafarers' rights because the inspections carried out on vessels calling at ports in countries that have ratified the Convention include inspections to ascertain whether seafarers' rights are protected.<sup>136</sup> As a result, seafarers' working onboard

129 'Each member shall implement its responsibilities under this Convention in such a way as to ensure that the ships that fly the flag of any State that has not ratified this Convention do not receive more favourable treatment

than the ships that fly the flag of any State that has ratified it.'

<sup>&</sup>lt;sup>130</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 25 May 2020).

<sup>&</sup>lt;sup>131</sup> Labour standards: Basic facts on the Maritime Labour Convention 2006. https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS 219665/lang-en/index.htm. (Accessed 22 September 2020).

<sup>&</sup>lt;sup>132</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 322.

<sup>&</sup>lt;sup>133</sup> "1. Every foreign ship calling, in the normal course of its business or for operational reasons, in the port of a Member may be the subject of inspection in accordance with paragraph 4 of Article V for the purpose of reviewing compliance with the requirements of this Convention (including seafarers' rights) relating to the working and living conditions of seafarers on the ship.

<sup>2.</sup> Each Member shall accept the maritime labour certificate and the declaration of maritime labour compliance required under Regulation 5.1.3 as prima facie evidence of compliance with the requirements of this Convention (including Seafarers' rights). Accordingly, the inspection in its ports shall, except in the circumstances specified in the Code, be limited to a review of the certificate and declaration.

<sup>3.</sup> Inspections in a port shall be carried out by authorised officers in accordance with the provisions of the Code and other applicable international arrangements governing port State control inspections in the Member. Any such inspection shall be limited to verifying that the matter inspected is in conformity with the relevant requirements set out in Articles and Regulations of this Convention and in Part A of the Code.

<sup>4.</sup>Inspections that may be carried out in accordance with this Regulation shall be based on an effective port State inspection and monitoring system to help ensure that the working and living conditions for seafarers on ships entering a port of the Member concerned meet the requirements of this Convention (including seafarers' rights).

<sup>5.</sup> Information about the system referred to in paragraph 4 of this Regulation, including the method used for assessing its effectiveness, shall be included in the Members reports pursuant to article 22 of the Constitution."

<sup>&</sup>lt;sup>134</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 25 May 2020).

<sup>&</sup>lt;sup>135</sup> See footnote 22.

<sup>&</sup>lt;sup>136</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 323.

vessels whose flag State have not ratified the Convention are protected when calling at ports that are member states of the Convention. 137

## 2.7 How the MLC is implemented

As an international instrument, the MLC relies on its member countries to implement its regulations through their national laws.<sup>138</sup> A country's' national laws and regulations would apply to seafarers, shipowners and vessels.<sup>139</sup> There are two key areas for flexibility in implementing the MLC; the first is a member state's ability to effect Part A of the Code's requirements through substantial equivalence.<sup>140</sup> The second area of flexibility is provided by framing the mandatory requirements in Part A of the Code in a more general manner, allowing a Member state to use its discretion.<sup>141</sup>

The MLC stipulates minimum standards that its member states must implement. The standards set by the MLC must be mirrored in the national standards of its member states and are subject to oversight by the Committee of Experts under the ILO supervisory system.<sup>142</sup>

Article IV, paragraph 5<sup>143</sup> of the MLC stipulates that seafarers' employment and their rights protected under the Convention may be implemented through national laws or regulations, collective bargaining agreements, through other measures or in practice, unless stipulated otherwise.<sup>144</sup> As a result, each member state is permitted to adopt the MLC's provisions in its law (as an Act of Parliament or Congress), in regulations

<sup>&</sup>lt;sup>137</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>138</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>139</sup> Labour standards: Basic facts on the Maritime Labour Convention 2006. https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS 219665/lang-en/index.htm. (Accessed 22 September 2020).

<sup>&</sup>lt;sup>140</sup>L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 302.

<sup>&</sup>lt;sup>141</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 302.

<sup>&</sup>lt;sup>142</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>143</sup> "Each Member shall ensure, within the limits of its jurisdiction, that the seafarers' employment and social rights set out in the preceding paragraphs of this Article are fully implemented in accordance with the requirements of this Convention. Unless specified otherwise in the Convention, such implementation may be achieved through national laws or regulations, through applicable collective bargaining agreements or through other measures or in practice."

<sup>&</sup>lt;sup>144</sup> Maritime Labour Convention, 2006 as amended, (2016) see Article IV Paragraph 5.

or through other subsidiary legislation.<sup>145</sup> In instances where the MLC does not require its measures to be implemented through legislation, a member state is permitted to allow certain matters to be dealt with through the use of collective bargaining agreements or alternative legal measures.<sup>146</sup> In some instances, a member state may elect not to take any further legal measures because a seafarer's rights that are protected under the MLC, are already adequately protected by the general law applied by a member state's national courts.<sup>147</sup>

Article VI, paragraph 3<sup>148</sup> and 4<sup>149</sup> of the MLC, stipulate that a national provision of a member state may be considered "substantially equivalent" to a provision contained in Part A of the Code if the member state "satisfies itself" that the legislation or measure "is conducive to the full achievement of the general object and purpose of the provision or provisions of Part A of the Code concerned" and "gives effect to the provision or provisions of Part A of the Code concerned."<sup>150</sup>

It is evident that the Member State's obligation is to "satisfy itself"; however, it is pertinent to note that this does not imply total autonomy of the member state because it is compulsory for the authorities responsible for monitoring implementation of the MLC at a national level to determine whether the required procedure of "satisfying themselves" has been carried out.<sup>151</sup> This responsibility includes determining whether the process of "satisfying themselves" has been carried out in good faith and in a manner that ensures the objectives of implementing the principles and rights set out

<sup>&</sup>lt;sup>145</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>146</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 301.

<sup>&</sup>lt;sup>147</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>148</sup> "A Member which is not in a position to implement the rights and principles in the manner set out in Part A of the Code may, unless expressly provided otherwise in this Convention, implement Part A through provisions in its laws and regulations or other measures which are substantially equivalent to the provisions of Part A."

<sup>&</sup>lt;sup>149</sup> "For the sole purpose of paragraph 3 of this Article, any law, regulation, collective agreement or other implementing measure shall be considered to be substantially equivalent, in the context of this Convention, if the Member satisfies itself that:

a) It is conducive to the full achievement of the general object and purpose of the provision or provisions of Part A of the Code concerned; and

b) It gives effect to the provision or provisions of Part A of the Code concerned."

<sup>&</sup>lt;sup>150</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>151</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 302.

in the MLC is adequately achieved in a way other than that indicated in Part A of the Code. 152

This is how member States should evaluate their national provisions when attempting to determine substantial equivalence, identifying the general purpose and object of Part A of the Code and determining whether or not the proposed national legislation could, in good faith, be considered as giving effect to Part A.<sup>153</sup> Any substantial equivalence that a Member State has adopted must consequently be stated in Part I of the Declaration of Maritime Labour Compliance, carried onboard vessels that have been certified.<sup>154</sup>

# 2.8 Advantages of the MLC

The MLC has improved the social dialogue at all levels and has ensured uniform compliance and verification of its implementation. As a result, it is advantageous to seafarers, shipowners and governments alike.

In the governments' case, the MLC has simplified their reporting obligations, requiring governments to conform to one Convention instead of several conventions. <sup>157</sup> Governments have more expansive powers of enforcement on all vessels, including foreign vessels calling at their ports and additional flexibility with the firmness of rights and implementing them. <sup>158</sup> This flexibility makes the Convention capable of broader ratification. <sup>159</sup>

\_

<sup>&</sup>lt;sup>152</sup> International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>153</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>154</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>155</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) IV.

<sup>&</sup>lt;sup>156</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>157</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>158</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 301.

<sup>&</sup>lt;sup>159</sup> Advantages of the Maritime Labour Convention, 2006. <a href="https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS">https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS</a> 153450/lang--en/index.htm. (Accessed 17 September 2020).

The MLC will ensure a more level playing field for shipowners to provide fair competition and marginalise substandard vessels' operation. The MLC prescribes minimum standards within the current industry practice, thereby making it easier for shipowners to comply with. Implementation of the MLC will ensure a more socially responsible industry and a more efficient and better-protected workforce. The MLC will also ensure that ships operate safely and securely with fewer delays in port.

In respect of seafarers, the MLC stipulates their fundamental employment rights in straightforward language and in one Convention; it also ensures seafarers are better informed of their rights and the remedies available to them when their rights are infringed. Seafarers have improved enforcement of their minimum working and living conditions and have the right to make complaints both onboard the vessel and ashore. 165

#### 2.9 Criticism of the MLC

Despite the MLC being a comprehensive Convention that protects the rights of seafarers, including those on-board vessels that are flagged in non-member states through the "no more favourable treatment clause", the MLC is not above criticism. In addition to being criticised about the guidelines contained in Part B which are not mandatory, and can be implemented at a lower standard, the MLC has been criticised for not addressing matters such as the protection of seafarers right to strike and the issuing of visas for shore leave.<sup>166</sup>

The crew change crisis of the Covid-19 Pandemic has illustrated the vulnerability of seafarers and how important their right to strike is. Since the Covid-19 Pandemic,

<sup>&</sup>lt;sup>160</sup> Advantages of the Maritime Labour Convention, 2006. <a href="https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS">https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS</a> 153450/lang--en/index.htm. (Accessed 17 September 2020).

<sup>&</sup>lt;sup>161</sup> L. Christodoulou-Varotsi and D.A Pentsov, Maritime Work Law Fundamentals: Responsible Shipowners, Reliable Seafarers. (2008) 315.

<sup>&</sup>lt;sup>162</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

Advantages of the Maritime Labour Convention, 2006. https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS 153450/lang--en/index.htm. (Accessed 17 September 2020).

<sup>&</sup>lt;sup>164</sup> International Labour Organisation. Maritime Labour Convention, 2006: Frequently asked questions. (2012) http://www.ilo.org/mlc. (Accessed: 31 May 2020).

<sup>&</sup>lt;sup>165</sup> Advantages of the Maritime Labour Convention, 2006. <a href="https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS">https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS</a> 153450/lang--en/index.htm. (Accessed 17 September 2020).

<sup>&</sup>lt;sup>166</sup>The Maritime Labour Convention: a review. *Lexology.* Available at

 $<sup>\</sup>frac{https://www.lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0}{December 2021).} \label{eq:lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0} (Accessed on 16 December 2021).$ 

many seafarers were stranded at sea due to the inability to facilitate crew changes. <sup>167</sup> As a result many international efforts were made to afford seafarers with some relief, this included a "call to action" to invoke the relevant sections of the MLC, however, the situation continued to deteriorate and reports from 2020 indicate that there were up to 300 000 seafarers stuck at sea. <sup>168</sup> The Covid-19 Pandemic has clearly illustrated the weaknesses in the labour regulatory regimes where workers are unable to take effective action at a workplace level. <sup>169</sup> Early criticism of the MLC highlighted the lack of provisions allowing seafarers to strike. Without some enabling provisions, the other rights offered by the MLC become meaningless. <sup>170</sup> Despite international input in the drafting of the MLC, seafarers do not have a work environment that allows them to take the type of action at shipboard level that would bring relief in a crisis of this nature. <sup>171</sup>

Nevertheless, the MLC is a great step forward in both consolidating the rights of seafarers and harmonising compliance and inspection procedures to be followed by flag states and port authorities.<sup>172</sup>

-

<sup>&</sup>lt;sup>167</sup> 'Why the MLC is failing as a regulatory response to the Covid-19 pandemic' *Splash*, 27 January 2021. Available at <a href="https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic">https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic</a> (Accessed 05 January 2022).

<sup>&</sup>lt;sup>168</sup>'Why the MLC is failing as a regulatory response to the Covid-19 pandemic' *Splash*, 27 January 2021. Available at <a href="https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic">https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic</a> (Accessed 05 January 2022).

<sup>&</sup>lt;sup>169</sup> 'Why the MLC is failing as a regulatory response to the Covid-19 pandemic' *Splash*, 27 January 2021. Available at <a href="https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic/">https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic/</a> (Accessed 05 January 2022).

<sup>&</sup>lt;sup>170</sup> 'Why the MLC is failing as a regulatory response to the Covid-19 pandemic' *Splash*, 27 January 2021. Available at <a href="https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic/">https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic/</a> (Accessed 05 January 2022).

<sup>&</sup>lt;sup>171</sup> 'Why the MLC is failing as a regulatory response to the Covid-19 pandemic' *Splash*, 27 January 2021. Available at <a href="https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic">https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic</a> (Accessed 05 January 2022).

<sup>&</sup>lt;sup>172</sup> The Maritime Labour Convention: a review. *Lexology*. Available at <a href="https://www.lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0">https://www.lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0</a> (Accessed on 16 December 2021).

## CHAPTER THREE: SEAFARERS' RIGHTS PRIOR TO THE MLC

The previous chapter examined the aims, development and implementation of shipowner liability under the MLC. In this chapter, the history of the national laws in the relevant three jurisdictions prior to implementing the MLC, will be analysed in order to determine the extent to which the law protected the rights of injured seafarers and where a shipowner's liability ended.

# 3.1 The United States of America

The seafarers' rights to recover damages for injuries have developed extensively from the initial remedies, restricting shipowner liability to maintenance, cure and unearned wages. The 1903 USA Supreme Court decision in *The Osceola* an example of one of the first times, a court declared that a seafarer could recover damages for injuries sustained due to a ship's unseaworthiness or a shipowners failure to provide equipment. The *Osceola* decision had far-reaching consequences and confirmed that a seafarer could not claim damages due to the negligence of the master or a member of the crew. The USA Congress later passed the Jones Act, which was ground-breaking legislation of its time, aimed at extending the rights of seafarers'. The One of the significant changes that the Jones Act to negligence of the employees of the shipowner, including the master of the vessel.

Prior to the enactment of the Jones Act, upon the request of the Secretaries of Labour and Commerce, Congress responded to *The Osceola* judgement by adding the following provision to the Seamen's Act of 1915:<sup>181</sup>

<sup>&</sup>lt;sup>173</sup> Louis J Gusmano, 'Seamen's Rights to Recover for injury against Either Shipowner or Charterer' (1980-1981) 55 Tul L Rev 1029.

<sup>&</sup>lt;sup>174</sup> The Osceola 198 U.S 158 (1903) at 175.

<sup>&</sup>lt;sup>175</sup> Louis J Gusmano, 'Seamen's Rights to Recover for injury against Either Shipowner or Charterer' (1980-1981) 55 Tul L Rev 1029.

 $<sup>^{\</sup>rm 176}$  The Osceola 198 U.S 158 (1903) at 175.

<sup>&</sup>lt;sup>177</sup> of 1920 also known as the Merchant Marine Act of 1920.

<sup>&</sup>lt;sup>178</sup> Louis J Gusmano, 'Seamen's Rights to Recover for injury against Either Shipowner or Charterer' (1980-1981) 55 Tul L Rev 1029.

<sup>&</sup>lt;sup>179</sup> of 1920.

<sup>&</sup>lt;sup>180</sup>George W Stumberg, 'The Jones Act. Remedies of Seamen' (1956) 17 Ohio State Law Journal 484.

<sup>&</sup>lt;sup>181</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 371.

"Provided, that in any suit to recover damages for any injury sustained on board vessel or in its service seamen having command shall not be considered fellow-servants with those under their authority."

Despite Congress' intention through this provision, to allow for seafarers to recover damages for injuries sustained due to the masters' negligence, <sup>182</sup> in *Chelentis v Luckenbach S.S Co*, <sup>183</sup> the court held that under the *Osceola*, a shipowners' duty was determined "without regard" to the relationship between a seafarer and other crewmembers. <sup>184</sup> Thereby rendering irrelevant, the Seaman's Act of 1915. <sup>185</sup>

As a result of the court's finding that the Seamen's Act<sup>186</sup> was irrelevant, the President of the International Seaman's Union of America testified before Congress to remedy the Act's nullity.<sup>187</sup> He stated that the court had found the Act irrelevant because it referred to the common law; however, the common law was not applicable onboard a vessel.<sup>188</sup> The President of the International Seaman's Union of America suggested that section 20 of the Seamen's Act be amended, Congress accepted this, and the amendment became Section 33 of the Jones Act of 1920.<sup>189</sup>

This amendment, leading to the enactment of the Jones Act,<sup>190</sup> extends the protection provided to railway employees under the Federal Employers' Liability Act<sup>191</sup> (FELA),

<sup>&</sup>lt;sup>182</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 376.

<sup>&</sup>lt;sup>183</sup> Chelentis v Luckenbach S.S Co, 247 U.S 372 (1918). The Court of Appeals and the District Court had come to the same conclusion.

<sup>&</sup>lt;sup>184</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 377.

<sup>&</sup>lt;sup>185</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 377.

<sup>&</sup>lt;sup>186</sup> of 1915.

<sup>&</sup>lt;sup>187</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 378.

<sup>&</sup>lt;sup>188</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 378.

<sup>&</sup>lt;sup>189</sup> "any seaman who shall suffer personal injury in the course of his employment may, at his election, maintain an action for damages at law, with the right of trial by jury, and in such action all statutes of the United States modifying of extending the common-law right or remedy in cases of personal injury to railway employees shall apply; and in case of the death of any seaman as a result of any such personal injury the personal representative of such seaman may maintain an action for damages at law with the right of trial by jury, and in such action all statutes of the United States conferring or regulating the right of action for death in the case of railway employees shall be applicable. Jurisdiction in such actions shall be under the court of the district in which the defendant employer resides or in which his principal office is located."

<sup>&</sup>lt;sup>191</sup> of 1908.

to seafarers.<sup>192</sup> Initially, seafarer who became ill or suffered an injury during their service were entitled to maintenance, cure and wages for the duration of the voyage.<sup>193</sup> Since the Jones Act did not change the pre-existing maritime law, an injured seafarer had two avenues he could pursue to recover damages.<sup>194</sup> If the seafarer suffered an injury due to the crew's negligence, the seafarer could recover damages under the Jones Act; however, if the seafarer sustained an injury due to unseaworthiness, he could recover damages under the traditional maritime law.<sup>195</sup>

Interpreting the Jones Act in this manner would substantially limit the intended effects of the Act. 196 A seafarer seeking damages under the Act, would be required to show negligence, causation and damages suffered, with a risk of damages being reduced if the seafarer was found to be negligent. In addition, the ability to recover damages under general maritime law would add little to the protection of seafarers' rights under the Jones Act. 197 The Jones Act covered unseaworthiness, 198 and while maintenance and cure remedies were essential, it offered less recovery of damages in some instances. 199 In addition, any claims that were brought under general admiralty law would be heard without the use of a jury. As a result, the ability of the Jones Act to further protect the interests of seafarers was criticised. 200 Because of this Congress ultimately improved upon and rewrote the Jones Act so that it could be used as a hybrid remedy, which added to the rights of seafarers related to unseaworthiness, maintenance and cure. 201

-

<sup>&</sup>lt;sup>192</sup> The Jones Act of 1920.

<sup>&</sup>lt;sup>193</sup> George W Stumberg, 'The Jones Act. Remedies of Seamen' (1956) 17 Ohio State Law Journal 484.

<sup>&</sup>lt;sup>194</sup> George W Stumberg, 'The Jones Act. Remedies of Seamen' (1956) 17 Ohio State Law Journal 484.

<sup>&</sup>lt;sup>195</sup> George W Stumberg, 'The Jones Act. Remedies of Seamen' (1956) 17 Ohio State Law Journal 484.

<sup>&</sup>lt;sup>196</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 381.

<sup>&</sup>lt;sup>197</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 381.

<sup>&</sup>lt;sup>198</sup> See TJ Schoenbaum. *Admiralty and Maritime Law.* 5<sup>th</sup> Hornbook ed. (2001) 297. FELA covered unseaworthiness to the extent that was required for the reasonable care in providing a safe vessel. The requirement of providing reasonable care in relation to a competent crew formed part of the common law obligation which FELA absorbed.

<sup>&</sup>lt;sup>199</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 381.

<sup>&</sup>lt;sup>200</sup> George W Stumberg, 'The Jones Act. Remedies of Seamen' (1956) 17 *Ohio State Law Journal* 489.

<sup>&</sup>lt;sup>201</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 382.

In *Panama R.R Johnson*,<sup>202</sup> the court rejected the notion that a seafarer could only sue 'at law' under the Jones Act and stated that a suit could be brought 'in admiralty'. In *Pacific S.S v Peterson*,<sup>203</sup> the court held that a seafarer could claim for maintenance and cure in conjunction with a claim for unseaworthiness or a claim under the Jones Act. In *Romero v International Terminal Operating Co*,<sup>204</sup> the court ultimately held that a claim under the Jones Act, unseaworthiness and maintenance and cure, could be decided in the same suit provided that the claimant did not have a double recovery.<sup>205</sup> Therefore it is proposed that the Jones Act, as well as the doctrine of unseaworthiness, maintenance and cure, can be regarded as a single source containing an injured seafarers' full remedies.<sup>206</sup>

Under this law, as a general rule, an injured seafarer could recover damages in the form of wages for the period of the voyage, a maintenance allowance for the outpatient and recovery period, costs of items such as medication, and hospitalisation and compensation for pain and suffering, loss of future earnings due to impairment and occasionally loss of wages.<sup>207</sup>

### 3.2 The United Kingdom

The protection of seafarers has always been a priority in maritime law, and admiralty courts have always exercised care to safeguard seafarers' interests. It has been established that a seafarer is entitled to maintenance and cure, his wages and passage back to port.<sup>208</sup> In England, the rights of a seafarer are protected through statute and case law.<sup>209</sup>

The regulatory measures to protect seafarers, developed in the 1960s due to years of concern regarding seafarers health and safety and the occasional action taken to

<sup>&</sup>lt;sup>202</sup> 264 US 375, AMC 551 (1924).

<sup>&</sup>lt;sup>203</sup> 264 US 375, AMC 1932 (1928).

<sup>&</sup>lt;sup>204</sup> 358 US 354, AMC 832 (1959).

<sup>&</sup>lt;sup>205</sup> Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce* 382.

<sup>&</sup>lt;sup>206</sup> Robert M Mallano, 'Seamen's Injuries: The Jones Act, Unseaworthiness, and Maintenance and Cure-The Siamese Triplets' (1963) *51 California Law Review* 412.

<sup>&</sup>lt;sup>207</sup> Max D Kossoris and Joseph Zisman, 'Workmen's Compensation for Seamen' (1946) *62 Monthly Labour Review* 851.

<sup>&</sup>lt;sup>208</sup> Fitz-Henry Smith, "Liability in the Admiralty for Injuries to Seamen." (1906) 19(6) Harvard Law Reviewl 419.

<sup>&</sup>lt;sup>209</sup> Fitz-Henry Smith, "Liability in the Admiralty for Injuries to Seamen." (1906) 19(6) Harvard Law Reviewl 423.

safeguard them.<sup>210</sup> By the medieval period, several codes that made provision for crew members' illness had been established; the code that directly affected English ships were the *Rolls of Oleron.*<sup>211</sup> Part seven of the *Rolls of Oleron* placed a significant duty on the master of a vessel to ensure the continued care and financial support of an ill or injured seafarer.<sup>212</sup>This duty is still present in current international conventions.<sup>213</sup>

While the *Rolls of Oleron* placed an inherent duty on the master of a vessel to ensure seafarer care, by 1867, several articles appeared in the British Medical Journal exposing the poor living conditions and neglect faced by seafarer.<sup>214</sup> This exposure ultimately led to the development of the *Merchant Shipping Act of 1867*,<sup>215</sup> which was the starting point to a cohesive approach to the health and care of British seafarers.<sup>216</sup> Therefore, the *Merchant Shipping Act of 1867* ultimately laid the foundation for many of the provisions used today to safeguard the health and safety of seafarers, ensuring that they can safely perform their duties.<sup>217</sup>

New developments in health care inevitably resulted in higher costs for shipowners; however, this also resulted in better health care for seafarers. Despite the improvement in healthcare provided to seafarers, shipowners viewed this as an additional cost. Shipowners overlooked the benefits of the improvements in health care and only implemented these developments if a failure to do so would result in the

-

<sup>&</sup>lt;sup>210</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>211</sup> The Rules of Oleron, c. 1266. <a href="https://www.admiraltylawguide.com/documents/oleron.html">www.admiraltylawguide.com/documents/oleron.html</a> (accessed 11 May 2020).

<sup>&</sup>lt;sup>212</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>213</sup> ILO Maritime Labour Convention 2006, Geneva; International Labour Conference. Standard A 4.1 Medical care on board ship and ashore.

<sup>&</sup>lt;sup>214</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>215</sup> Section 4(2) stated that ships carry medical stores in a specified scale and an approved guide to the use of the medical stores. Section 5 placed a duty on suppliers of medical stores to ensure their quality and placed penalties on suppliers. Section 7 placed liability on the shipowner or master to pay wages for up to three months and for the cost of treating any illnesses that arises from their neglect to provide proper food, water and medicine.

<sup>&</sup>lt;sup>216</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>217</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>218</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

shipowner being criticised.<sup>219</sup> While the *Merchant Shipping Act of 1867* protected seafarers, ensuring their health care and safety at sea, the Act did not extend this protection to seafarers while they were ashore.<sup>220</sup> As a result, seafarers health and safety while ashore became a pressing issue that needed to be addressed.<sup>221</sup>

After World War I (WW1), the need for international collaboration was at the fore; seafarers were recognised as the most international group of workers, as a result bodies such as the ILO focused their early work on the wellbeing of workers in the maritime sector.<sup>222</sup>

Admiralty law intrinsically has its own rights and remedies applicable within a particular area that cannot be utilised outside its operation area.<sup>223</sup> An example of this is the maritime lien and the action in rem.<sup>224</sup> Statutes are the principal legal source in maritime law; statues also play an instrumental role in judicial development and English law.<sup>225</sup> Court integration and statutes have had a significant effect on resolving conflicts. However, English "maritime law" is rooted in statutes, procedural rules and the judicial doctrine of Admiralty, common law and equity.<sup>226</sup> The consolidation in *The Merchant Shipping Act 1995* of the preceding Merchant Shipping Acts goes far to provide a code of substantive rules; however, it does not refer to maritime issues that appear in other statutes.<sup>227</sup>

The recent developments in English law, particularly *The Merchant Shipping Act of 1995*, have their roots in International Conventions.<sup>228</sup> Some of these developments are because of the United Kingdoms' membership with the European Union, the United Nations Conference on Trade and Development (UNCTAD), the International

<sup>&</sup>lt;sup>219</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>220</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>221</sup> A Kennerley. *British Seamen's Missions and Sailors' Homes 1815-1970: voluntary welfare provision for serving seafarers.* PhD thesis. University of Plymouth, (1989). 102-104.

<sup>&</sup>lt;sup>222</sup> Tim Carter. *Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain.* (2014).

<sup>&</sup>lt;sup>223</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 221.

<sup>&</sup>lt;sup>224</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 221.

<sup>&</sup>lt;sup>225</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 222.

<sup>&</sup>lt;sup>226</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 223.

<sup>&</sup>lt;sup>227</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 224.

<sup>&</sup>lt;sup>228</sup> The Rules of Oleron, United Nations Convention on the Law of the Sea 1982, International Convention for the Safety of Life at Sea (SOLAS) 1974.

Maritime Organisation (IMO) and the International Labour Organisation (ILO).<sup>229</sup> The law of the European Union governs its member states' laws; therefore, regulations, directives, and decisions made by the Union will be binding on its member states.<sup>230</sup> In the United Kingdom, international regulations, directives and decisions are implemented through domestic legislation, such as *The Merchant Shipping Act of* 1995.<sup>231</sup>

Before implementing the MLC in the UK, an injured seafarers' rights and right of recourse against a shipowner was consolidated in *The Merchant Shipping Act of 1995*. The Act protects seafarers by ensuring they are paid wages monthly or in full when they leave the ship.<sup>232</sup> A seafarers wages must be paid when a UK ship has been wrecked or lost, thereby terminating employment before the termination date or where the vessel has been sold and is no longer a UK ship, thereby terminating employment before the agreed date.<sup>233</sup> Minimum requirements for crew accommodation are stipulated to ensure satisfactory living conditions.<sup>234</sup> Seafarers are entitled to medical treatment at the shipowners' expense while onboard the vessel and ashore while outside the UK.<sup>235</sup> If a seafarer dies at sea or ashore outside the UK, the shipowner will be liable for the deceased's costs of the burial or cremation.<sup>236</sup> Seafarers are entitled to relief, maintenance and return to the UK at the end of a voyage or in the instance of a shipwreck.<sup>237</sup>

#### 3.3 South Africa

South African admiralty law has a strong link to English law; therefore, South African admiralty law can never be considered without reference to the history of admiralty law in the UK.<sup>238</sup> South African admiralty law is a hybrid system that comprises statute

<sup>&</sup>lt;sup>229</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 223.

<sup>&</sup>lt;sup>230</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 224.

<sup>&</sup>lt;sup>231</sup> DC Jackson. *Enforcement of Maritime Claims*. 4<sup>th</sup> ed. (2005) 225.

<sup>&</sup>lt;sup>232</sup> Section 30 (1) (a) and (b).

<sup>&</sup>lt;sup>233</sup> Section 38 (1) and (2).

<sup>&</sup>lt;sup>234</sup> Section 43.

<sup>&</sup>lt;sup>235</sup> Section 45.

<sup>&</sup>lt;sup>236</sup> Section 45 (2) and (3).

<sup>&</sup>lt;sup>237</sup> Section 73.

<sup>&</sup>lt;sup>238</sup> s6 of the Admiralty Jurisdiction and Regulation Act 105 of 1983 (AJRA) obliges a South African court, when faced with most marine matters to apply English law as at 1983. See *Transol Bunker v MV Andrico Unity and Others* 1989 (4) SA 325 (A) at 334H-336C.

and common law.<sup>239</sup> The South African system of law is precedent based whereby courts pronounce on issues by interpreting and enforcing statutes and developing the common law.<sup>240</sup>

The origins of South African shipping law can be traced back to the laws of the first European settlers in the Cape in 1652.<sup>241</sup> The Dutch settlement inevitably resulted in the application of Dutch law, which was applied under the Dutch-Estates General's authority.<sup>242</sup> Dutch occupation of the Cape continued until 1806 when the British occupied the Cape during this period, the law in South Africa falling under Dutch rule developed alongside the law in Europe.<sup>243</sup>

South African common law is a combination of Roman-Dutch and English law; the British established the court system, and British judges presided over matters.<sup>244</sup> British rule brought colonial status to the Cape in 1815, and in 1827 the Cape Supreme Court was established with three British Judges being appointed.<sup>245</sup> English practices and admiralty law was introduced to the South African legal system through the *Colonial Courts of Admiralty Act 1890*, which gave every court within British possession admiralty jurisdiction.<sup>246</sup>

Before enacting The Admiralty Jurisdiction Regulation Act (AJRA), English statutes relating to admiralty matters as of 1891 were applicable in South Africa; therefore,

Common Law in South Africa (1996) 33-64.

<sup>&</sup>lt;sup>239</sup> A Practical Overview on Enforcement and Defence of Maritime Claims in South Africa: Including Casualties and Carriage of Goods by Sea. (2014) *Shepstone & Wylie Attorneys*. <a href="http://wylie.co.za/wp-">http://wylie.co.za/wp-</a>

<sup>&</sup>lt;u>content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf</u>. (Accessed 30 May 2020). <sup>240</sup> A Practical Overview on Enforcement and Defence of Maritime Claims in South Africa: Including Casualties

and Carriage of Goods by Sea. (2014) *Shepstone & Wylie Attorneys*. <a href="http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf">http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf</a>. (Accessed 30 May 2020).

<sup>&</sup>lt;sup>241</sup> Origins of Admiralty Jurisdiction & Shipping Law in South Africa: The Anglo -Dutch juxtaposition- common law in a civilian system. (2009) *Juta and Company (Pty) Ltd.* 

http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/7/8?f=templates\$fn=default.htm. (Accessed 10 June 2020). <sup>242</sup> See Fagan, *Roman-Dutch Law in its South African Historical Context* in *Southern Cross- Civil law and* 

<sup>&</sup>lt;sup>243</sup> Origins of Admiralty Jurisdiction & Shipping Law in South Africa: The Anglo -Dutch juxtaposition- common law in a civilian system. (2009) *Juta and Company (Pty) Ltd.* 

http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/7/8?f=templates\$fn=default.htm. (Accessed 10 June 2020).

244 A Practical Overview on Enforcement and Defence of Maritime Claims in South Africa: Including Casualties and Carriage of Goods by Sea. (2014) Sheptone 8. Mylie Attorneys, http://wwlie.co.za/wp-

and Carriage of Goods by Sea. (2014) *Shepstone & Wylie Attorneys*. <a href="http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf">http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf</a>. (Accessed 30 May 2020).

<sup>&</sup>lt;sup>245</sup> Origins of Admiralty Jurisdiction & Shipping Law in South Africa: The Anglo -Dutch juxtaposition- common law in a civilian system. (2009) *Juta and Company (Pty) Ltd.* 

http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/7/8?f=templates\$fn=default.htm. (Accessed 10 June 2020). <sup>246</sup> A Practical Overview on Enforcement and Defence of Maritime Claims in South Africa: Including Casualties and Carriage of Goods by Sea. (2014) *Shepstone & Wylie Attorneys*. <a href="http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf">http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf</a>. (Accessed 30 May 2020).

South African courts were to apply English admiralty law and precedents.<sup>247</sup> The enactment of AJRA has broadened the ambit of South African courts as it has introduced the application of English common law, which was in force up till November 1983.<sup>248</sup>

The laws applicable to the employment of the master and crew are a combination of private rights and obligations and statutory controls.<sup>249</sup>In South Africa today, the High Court exercises its admiralty jurisdiction when hearing claims of crew members.<sup>250</sup> While the High Court exercising its admiralty jurisdiction has the authority to hear these claims, the law that is to be applied to the claims is not as straight forward.<sup>251</sup> In 1891 the English admiralty court had jurisdiction:

"Over any claim by a seaman of any ship for wages owed by him on board the ship, whether the same be due under a special contract or otherwise, and also over any claim by the master of any ship for wages earned by him on board the ship, and for disbursements made by him on account of the ship" <sup>252</sup>.

The jurisdiction of the 1861 *Admiralty Court Act* applied only to the collection of wages.<sup>253</sup> Generally, matters that were not related to wage claims and, in some instances, complaints of ill-treatment were heard by the common law courts.<sup>254</sup>The application of s6 of The Admiralty Jurisdiction Regulation Act (AJRA)<sup>255</sup> would result in wage claims in South Africa being subject to English law as it were in 1983; however,

\_

<sup>&</sup>lt;sup>247</sup> A Practical Overview on Enforcement and Defence of Maritime Claims in South Africa: Including Casualties and Carriage of Goods by Sea. (2014) *Shepstone & Wylie Attorneys*. <a href="http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf">http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf</a>. (Accessed 30 May 2020). <sup>248</sup> A Practical Overview on Enforcement and Defence of Maritime Claims in South Africa: Including Casualties and Carriage of Goods by Sea. (2014) *Shepstone & Wylie Attorneys*. <a href="http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf">http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf</a>. (Accessed 30 May 2020). <sup>249</sup> Chapter 5, Master and Crew: Jurisdiction and applicable law. (2009) *Juta and Company (Pty) Ltd*. <a href="http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020). <sup>250</sup> Admiralty Jurisdiction Regulation Act, s1 (1)(s).

<sup>&</sup>lt;sup>251</sup> Chapter 5, Master and Crew: Jurisdiction and applicable law. (2009) *Juta and Company (Pty) Ltd.*<a href="http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020).

<sup>252</sup> Admiralty Court Act, 1861, s10. The word 'seaman' was interpreted at the time to include all paid persons on board, other than the Master. Williams and Bruce, *Jurisdiction and Practice of the English Courts in Admiralty Actions and Appeals.* (1986) 201.

<sup>&</sup>lt;sup>253</sup> Chapter 5, Master and Crew: Jurisdiction and applicable law. (2009) *Juta and Company (Pty) Ltd.* <a href="http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020). <a href="http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020). <a href="https://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">https://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020). <a href="https://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">https://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020).

other claims relating to employment would be subject to South African Roman-Dutch law.<sup>256</sup>

Therefore, a seafarer right to wages would have been protected under English law and applied in South Africa. Any other rights and claims of a seafarer would be protected in South African law under the *Merchant Shipping Act.*<sup>257</sup> The Act ensures that a seafarer is entitled to his wages and that he will be paid the full amount or the balance thereof upon discharge of the vessel.<sup>258</sup> If a seafarers services are terminated before the contemplated date, through wreck<sup>259</sup> or if he has been left behind at a foreign port due to illness or injury,<sup>260</sup> he will be entitled to his wages until he has arrived at the agreed port.<sup>261</sup> Should a seafarer be left behind at a foreign port, he is entitled to his wages and repatriation expenses.<sup>262</sup> The seafarer will be entitled to his maintenance, the provision of necessaries such as clothing and all costs necessary to return him to the intended port.<sup>263</sup> A seafarer is entitled to proper accommodation.<sup>264</sup> The master of the vessel must provide a seafarer with bedding, towels and any other articles for personal use that may be required.<sup>265</sup> A seafarer is entitled to medical care on board the vessel; the master or shipowner must ensure an adequate supply of medication and equipment to treat any illness or injury.<sup>266</sup>

Each country has its own statutes containing and protecting the rights of seafarers. However, due to the international nature of shipping and the early developments of doctrine such as the *Rolls of Oleron*, it is evident that seafarers' basic protection covers their right to wages, health care on board the vessel, maintenance and cure.

<sup>&</sup>lt;sup>256</sup> Chapter 5, Master and Crew: Jurisdiction and applicable law. (2009) *Juta and Company (Pty) Ltd.* <a href="http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020). <sup>257</sup> Act 57 of 1951.

<sup>&</sup>lt;sup>258</sup> Section 120 amended be s19 of Act 30 of 1959 and substituted by s12 of Act 42 of 1969.

<sup>&</sup>lt;sup>259</sup> Section 140 (1) (a) substituted by s5 (a) of Act 3 of 1981.

<sup>&</sup>lt;sup>260</sup> Section 140 (1) (b) amended by s21 (a) of Act 30 of 1959.

<sup>&</sup>lt;sup>261</sup> Section 140 (1).

<sup>&</sup>lt;sup>262</sup> Section 154.

<sup>&</sup>lt;sup>263</sup> Section 154 (b).

<sup>&</sup>lt;sup>264</sup> Section 161.

<sup>&</sup>lt;sup>265</sup> Section 160.

<sup>&</sup>lt;sup>266</sup> Section 167.

# CHAPTER 4: SEAFARER'S RIGHTS AFTER THE IMPLEMENTATION OF THE MLC

The preceding chapters examined the MLC, its development, aims, implementation, and the history of the laws protecting seafarers in the respective jurisdictions. These chapters provide a clear indication of the level of protection afforded to seafarers prior to the introduction of the MLC. This chapter will examine the laws in the respective jurisdictions after the introduction of the MLC in order to assess the extent to which the MLC has been implemented and whether it has been effective in protecting seafarers' rights.

#### 4.1 The United States of America

Despite being at the forefront of seafarer protection through the enactment of regulations such as the Jones Act, <sup>267</sup> the USA has experienced difficulty protecting the rights of foreign seafarers entering its ports. <sup>268</sup> However, by ratifying the MLC, <sup>269</sup> the USA will be in a better position to protect foreign seafarers' rights and promote seafarer protection of vessels that call at USA ports. Relying on the powers conferred by the MLC to its member states, the USA would be able to inspect foreign flagged vessels of other member states, to ensure compliance with MLC provisions including fair living and working conditions of seafarers manning such vessels. <sup>270</sup>

Ports and waterways in the USA are heavily occupied by USA flagged vessels and American seafarers due to the requirement that only USA flagged vessels may travel within its domestic ports.<sup>271</sup> The Jones Act has significantly protected American seafarers within the USA, however, USA-flagged merchant vessels only make up 2% of the global cargo tonnage.<sup>272</sup> In light of the USA not ratifying the MLC, the Jones Act currently only protects American seafarers because USA Port State Control cannot

<sup>&</sup>lt;sup>267</sup> The Jones Act- the Foundation of the Merchant Marine. *American Maritime Congress.* (2012). http://www.americanmaritime.org/about/jonesact.pdf. (Accessed 5 May 2020).

<sup>&</sup>lt;sup>268</sup> "ILO Convention No. 147, Student Guide" (2010) *Int'l Labour Org.* 13.

<sup>&</sup>lt;sup>269</sup> As of 16 February 2021, the United States of America has not ratified the MLC,

http://skuld.com/topics/people/mlc-2006/ratification-of-mlc-convention/. (Accessed 15 March 2021).

<sup>&</sup>lt;sup>270</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 *Hofstra Lab& Emp LJ*. 168.

<sup>&</sup>lt;sup>271</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 *Hofstra Lab& Emp LJ*. 168.

<sup>&</sup>lt;sup>272</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 173.

inspect and hold foreign flagged vessels to the standards prescribed in the MLC.<sup>273</sup> At face value, it may appear that ratifying the MLC would not be of significant value to the USA because of the protection already offered in the Jones Act. However, there are certain notable benefits to the USA ratifying the MLC.<sup>274</sup>

The first consideration of value to stakeholders in the USA, would be the acquisition of certificates or declarations, in compliance with the inspection protocols set out in the MLC. Were these within the ambit of the MLC, USA flagged vessels could relatively easily obtain MLC compliance certificates due to these vessels already having complied with the requirements set out in the Jones Act and general labour laws applicable within the USA. Which are highly compatible with the provisions of the MLC.<sup>275</sup> However, failing to ratify the MLC, would subject USA flagged vessels that are not in possession of these compliance certificates, to enhanced scrutiny and inspection by other MLC member port states.<sup>276</sup>

While there is limited risk of any significant non-compliance being detected on USA flagged vessels within other port states, these in-depth inspections leave less time for seafarers to go ashore and enjoy recreational breaks,<sup>277</sup> especially when a vessel is only in port for twelve to twenty-four hours. Being MLC compliant would mean less inspections, allowing seafarers who spend long periods at sea with little social interaction, more time to go ashore and break the monotony of life at sea.<sup>278</sup>

The second notable consideration would be the future of American seafarers on foreign flagged vessels.<sup>279</sup> While it is rare to find American seafarers on foreign-flagged vessels, policies have been proposed to the Senate, which could make this a

<sup>&</sup>lt;sup>273</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 173.

<sup>&</sup>lt;sup>274</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 197.

<sup>&</sup>lt;sup>275</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 197.

<sup>&</sup>lt;sup>276</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 197.

<sup>&</sup>lt;sup>277</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 198.

<sup>&</sup>lt;sup>278</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 198.

<sup>&</sup>lt;sup>279</sup>P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 198.

more frequent future occurrence.<sup>280</sup> Currently under the Jones Act only vessels owned by American citizens, operating with a predominantly American crew and built in the USA, can operate within USA domestic ports.<sup>281</sup> The late Senator John McCain proposed changes to the Jones Act extending its protective measures to seafarers on foreign flagged vessels and allowing such vessels to operate between USA ports. The amendments to the Jones Act submitted to the Senate, ultimately failed because these amendments would end the preferential treatment of vessels built in the USA, severely prejudicing the shipbuilding industry and seafarers employed within the USA.<sup>282</sup> Had the proposed amendments been successful, USA seafarers could have found themselves in a position where they were forced to work on foreign-flagged vessels.<sup>283</sup> By adopting the MLC, the United States Port State Control would be in a position to carry out compliance inspections and hold foreign-flagged vessels accountable to the standard of labour conditions set out in the MLC, thus protecting American seafarers on board foreign-flagged vessels.

The most recent ILO Convention that the United States has ratified is Convention no 147, The Merchant Shipping (Minimum Standards) Convention. <sup>284</sup> The purpose of the Convention is to improve the conditions of employment on merchant vessels and allow the Port States that are parties to the Convention, to take the required actions necessary to protect the health and safety of seafarers manning merchant vessels calling at those ports. <sup>285</sup> The Convention aims to introduce minimum labour standards throughout the shipping industry, implementing these standards through practical Port State Control exam and reporting systems. <sup>286</sup>

\_\_\_

<sup>&</sup>lt;sup>280</sup> See Michael Finch II, Lawmakers: Repealing Jones Act Would Have a 'Harmful Effect on our Economy," <a href="http://www.al.com/business/index.ssf/2015/01/lawmakers">http://www.al.com/business/index.ssf/2015/01/lawmakers</a> oppose mccain jones act repeal amendment k <a href="https://exessed.25">eystone.html</a>. (Accessed 25 June 2020).

<sup>&</sup>lt;sup>281</sup> See Michael Finch II, Lawmakers: Repealing Jones Act Would Have a 'Harmful Effect on our Economy," <a href="http://www.al.com/business/index.ssf/2015/01/lawmakers">http://www.al.com/business/index.ssf/2015/01/lawmakers</a> oppose mccain jones act repeal amendment k <a href="https://www.al.com/business/index.ssf/2015/01/lawmakers">eystone.html</a>. (Accessed 25 June 2020).

<sup>&</sup>lt;sup>282</sup> See *McCain's Job Killing Plan to Repeal Jones Act Fails*. (showing that 400,000 U.S shipbuilding, seafaring, and supply chain jobs could potentially be lost in the maritime industry).

<sup>&</sup>lt;sup>283</sup> See *McCain's Job Killing Plan to Repeal Jones Act Fails*. (showing that 400,000 U.S shipbuilding, seafaring, and supply chain jobs could potentially be lost in the maritime industry).

<sup>&</sup>lt;sup>284</sup> The United States of America ratified the Convention on the 15 June 1988, and it is currently still in force. https://www.ilo.org/dyn/normlex/en/f?p=1000:11300:0::NO:11300:P11300 INSTRUMENT ID:312292. (Accessed 10 July 2020).

<sup>&</sup>lt;sup>285</sup> https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:12100:0::NO::P12100\_ILO\_CODE:C147. (Accessed 10 July 2020).

<sup>&</sup>lt;sup>286</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 201.

In theory, Convention no 147 was envisioned to advance seafarers' human and labour rights; however, the Convention has fallen short in its applicability in the United States of America.<sup>287</sup> This is because several of the ILO conventions have been ratified and implemented sporadically by various nations, resulting in confusion.<sup>288</sup> Sadly, Convention no 147 has been no different from preceding conventions, as only 33 ILO member states have ratified it.<sup>289</sup> In terms of the provisions of Convention no 147, a "clearly hazardous condition" must be present, permitting the USA to detain a vessel of another member state.<sup>290</sup> For the United States Coast Guard (USCG) to have the authority to intervene, the hazardous condition "must be related to navigation, vessel safety and the protection of the marine environment". This means there is nothing concerning labour standards that would allow the USCG to intervene.<sup>291</sup> This clearly indicates how difficult it is for Port State Control (PSC) in the United States of America to address substandard labour conditions in foreign-flagged vessels calling at their ports.<sup>292</sup> Adopting the MLC would create a favourable platform for PSC within the USA to enforce acceptable labour standards ensuring that crew members on board vessels that they inspect, are competent, well-trained and rested.<sup>293</sup>

Ratifying the MLC, would enable the USA to further its commitment to addressing labour deficiencies by adding this as a criterion to their list of inspections, thus clamping down on flag states and vessels that are non-compliant with basic labour standards provided for in the MLC. <sup>294</sup> This would inevitably enhance the labour rights

\_

<sup>&</sup>lt;sup>287</sup> See US Coast Guard, No 16711.12A, Commandment Instruction (1996). https://www.uscg.mil/directives/ci/16000-16999.Cl 16711 12A.pdf. (Accessed 8 August 2020).

<sup>&</sup>lt;sup>288</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 201.

<sup>&</sup>lt;sup>289</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 201.

<sup>&</sup>lt;sup>290</sup>Article 4 (1) states: "If a Member which has ratified this Convention and in whose port a ship calls in the normal course of its business or for operational reasons receives a complaint or obtains evidence that the ship does not conform to the standards of this Convention, after it has come into force, it may prepare a report addressed to the government of the country in which the ship is registered, with a copy to the Director-General of the International Labour Office, and may take measures necessary to rectify any conditions on board which are clearly hazardous to safety or health." See P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 201.

<sup>&</sup>lt;sup>291</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 201.

<sup>&</sup>lt;sup>292</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 202.

<sup>&</sup>lt;sup>293</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 202.

<sup>&</sup>lt;sup>294</sup> P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 204.

of seafarers within the maritime industry, while encouraging flag states to play an active role in promoting seafarer protection globally. <sup>295</sup>

#### 4.2 The United Kingdom

The maritime industry plays an integral role in the economy of the United Kingdom. As of December 2019, it was ranked 24<sup>th</sup> in the world fleet tonnage statistics.<sup>296</sup> The UK's maritime industry contributes around £14,5 billion to the country's economy annually and provides approximately 185 000 jobs to land-based employees and seafarers.<sup>297</sup>

England and Wales have a common law framework that consists of a combination of case law and legislation. In particular, shipping law has historically been developed by decided cases.<sup>298</sup> The Merchant Shipping Act 1995, which was reformed in 1995, is a significant piece of legislation, and various statutory instruments have been enacted under it.<sup>299</sup>

The UK has currently ratified all major international maritime conventions, which are generally implemented through its domestic legislation.<sup>300</sup> While the UK was a member of the European Union, the regulations and directives made by the European Union had an indirect or direct effect on the operations of England and Wales.<sup>301</sup> Following the triggering of Article 50 in March 2017, the UK left the European Union in January 2020 and went into a period of transition which ended on 31 December 2020.<sup>302</sup>

<sup>&</sup>lt;sup>295</sup>P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 Hofstra Lab& Emp LJ. 204.

<sup>&</sup>lt;sup>296</sup> George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020) <a href="https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales">https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales</a>. (Accessed 10 January 2021).

<sup>&</sup>lt;sup>297</sup> George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020) <a href="https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales">https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales</a>. (Accessed 10 January 2021).

<sup>&</sup>lt;sup>298</sup> George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020) <a href="https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales">https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales</a>. (Accessed 10 January 2021).

<sup>&</sup>lt;sup>299</sup> George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020) <a href="https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales">https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales</a>. (Accessed 10 January 2021).

<sup>&</sup>lt;sup>300</sup> DC Jackson. *Enforcement of Maritime Claims*. Fourth Edition (2005). 349.

<sup>&</sup>lt;sup>301</sup> George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020) <a href="https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales">https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales</a>. (Accessed 10 January 2021).

<sup>&</sup>lt;sup>302</sup> George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020) <a href="https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales">https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales</a>. (Accessed 10 January 2021).

During the transition period, the UK continued to operate under the European Union regulations. However, after the 31st of December 2020, these regulations will not have effect unless they are expressly implemented into UK law.<sup>303</sup> Although the UK's domestic legislation will still be in place to implement the directives, it will need to be updated to address any gaps that may have emerged.<sup>304</sup> The MLC entered into force in the UK on the 14th of August 2014, with the UK being the 41st ILO member state to ratify the MLC.<sup>305</sup>

In the UK, the Maritime and Coastguard Agency (MCA) works to prevent the loss of life on the coast and at sea.<sup>306</sup> The MCA produces legislation and guidance on maritime matters as well as provides certification to seafarers.<sup>307</sup> The MCA is an executive agency that the Department of Transport sponsors.<sup>308</sup>

In the UK,<sup>309</sup> the MLC does not apply to seafarers serving on vessels navigating inland or in sheltered waters that are subject to port regulations, such as fishing vessels, warships, naval auxiliaries and traditional ships such as dhows.<sup>310</sup>This is because the MLC aims to regulate rights of seafarers in merchant vessels that operate internationally. This allows states to exclusively legislate vessels that operate within their ports.

#### 4.2.1 Crew Agreements

The Convention, enacted in domestic legislation, provides that seafarers have employment contracts or crew agreements entered into between themselves and shipowners.<sup>311</sup> The crew agreement must be made between each person employed

<sup>&</sup>lt;sup>303</sup> George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020) <a href="https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales">https://thelawreviews.co.uk/title/the-shipping-law-review/united-kingdom-england--wales</a>. (Accessed 10 January 2021).

<sup>&</sup>lt;sup>304</sup> DC Jackson. *Enforcement of Maritime Claims*. Fourth Edition (2005). 349.

<sup>&</sup>lt;sup>305</sup> https://www.ilo.org/dyn/normlex/en/f?p=1000:11200:0::NO:11200:P11200 COUNTRY ID:102651 (Accessed 18 February 2021).

<sup>&</sup>lt;sup>306</sup> https://www.gov.uk/government/organisations/maritime-and-coastguard-agency (Accessed 20 February 2021)

<sup>&</sup>lt;sup>307</sup> <a href="https://www.gov.uk/government/organisations/maritime-and-coastguard-agency">https://www.gov.uk/government/organisations/maritime-and-coastguard-agency</a> (Accessed 20 February 2021)

<sup>308</sup> https://www.gov.uk/government/organisations/maritime-and-coastguard-agency (Accessed 20 February 2021)

<sup>&</sup>lt;sup>309</sup> The Merchant Shipping Act 1995, Section 24.

<sup>&</sup>lt;sup>310</sup> Seafarer working and living rights: Maritime Labour Convention- Gov.UK, <a href="https://www.gov.uk/seafarer-working-and-living-rights/maritime-labour-convention">https://www.gov.uk/seafarer-working-and-living-rights/maritime-labour-convention</a> (Accessed 6 March 2020).

<sup>&</sup>lt;sup>311</sup> Seafarer working and living rights: Maritime Labour Convention- Gov.UK, <a href="https://www.gov.uk/seafarer-working-and-living-rights/maritime-labour-convention">https://www.gov.uk/seafarer-working-and-living-rights/maritime-labour-convention</a> (Accessed 6 March 2020).

as a seafarer on a UK vessel and the person employing him or her.<sup>312</sup> Crew agreements will contain all information relating to the seafarer's employment, such as wages,<sup>313</sup> minimum age requirements,<sup>314</sup> and crew discharge.<sup>315</sup> Crew agreements must be in writing and signed by both parties,<sup>316</sup> and a copy of this signed agreement must be kept on board the vessel to which it relates when the ship goes out to sea.<sup>317</sup>

The UK has passed legislation that will implement amendments to the application of the Maritime Labour Convention to all UK flagged vessels.<sup>318</sup> The new legislation to be implemented will protect the rights of seafarers caught up in piracy and other criminal acts onboard the vessel.<sup>319</sup> This legislation provides that if a seafarer is held captive on board a vessel, he/she is still entitled to be remunerated by the shipowner, and their employment contract remains in place.<sup>320</sup>

## 4.2.2 Wages

The UK Merchant Shipping Act provides that seafarers employed under a crew agreement must be paid their basic wages in specified intervals not exceeding a month.<sup>321</sup> Seafarers are further entitled to the full payment of any outstanding wages owed to them upon their discharge from the vessel.<sup>322</sup> The master of every UK flagged vessel must deliver a statement of account to every seafarer employed under a crew agreement; the statement of account must state the wages due to the seafarer as well as any deductions.<sup>323</sup> In the event of a UK flagged vessel being wrecked or is lost at sea, resulting in the termination of the seafarer's employment before the date

<sup>&</sup>lt;sup>312</sup> The Merchant Shipping Act 1995, Section 25 (1).

<sup>&</sup>lt;sup>313</sup> The Merchant Shipping Act 1995, Section 30 (1).

<sup>&</sup>lt;sup>314</sup> The Merchant Shipping Act 1995, Section 55.

<sup>&</sup>lt;sup>315</sup> The Merchant Shipping Act, 1995, Section 27 (2).

<sup>&</sup>lt;sup>316</sup> The Merchant Shipping Act, 1995, Section 25 (1).

<sup>&</sup>lt;sup>317</sup> The Merchant Shipping Act, 1995 Section 25 (4).

<sup>&</sup>lt;sup>318</sup> The UK has now passed legislation which will implement amendments to the Maritime Labour Convention. This will affect all ships that enter the UK.

<sup>&</sup>lt;sup>319</sup> UK Law Ensures Continued Employment Rights for Seafarers Held Captive. *The Maritime Executive*. 2 November 2020. <a href="https://www.maritime-executive.com/corporate/uk-law-ensures-continued-employment-rights-for-seafarers-held-captive">https://www.maritime-executive.com/corporate/uk-law-ensures-continued-employment-rights-for-seafarers-held-captive</a> (Accessed 16 February 2021).

<sup>&</sup>lt;sup>320</sup> UK Law Ensures Continued Employment Rights for Seafarers Held Captive. *The Maritime Executive*. 2 November 2020. <a href="https://www.maritime-executive.com/corporate/uk-law-ensures-continued-employment-rights-for-seafarers-held-captive">https://www.maritime-executive.com/corporate/uk-law-ensures-continued-employment-rights-for-seafarers-held-captive</a> (Accessed 16 February 2021).

<sup>321</sup> The Merchant Shipping Act, 1995 Section 30 (4).

<sup>&</sup>lt;sup>322</sup> The Merchant Shipping Act, 1995 Section 30 (1).

<sup>323</sup> The Merchant Shipping Act, 1995 Section 31 (1).

contemplated in the crew agreement, the seafarer will be entitled to two months wages, commencing from the date of the wreck.<sup>324</sup>

#### 4.2.3 Obligation of the shipowner in respect of seaworthiness

In every contract of employment or crew agreement concluded between the owner of a UK flagged vessel and a seafarer employed on that vessel, there is an implied obligation that the owner, master, and any agent charged with loading, preparing the ship or sending the ship out to sea, shall use all reasonable means to ensure that the vessel is seaworthy at the time that the voyage commences and for the duration of the voyage. This obligation imposed on a shipowner applies notwithstanding any agreement to the contrary. This offers significant protection to seafarers and ensures their wellbeing because a shipowner cannot add clauses into the crew agreement that would allow him/her to contract out of their obligation of ensuring that the vessel is seaworthy.

#### 4.2.4 Crew Accommodation

Depending on the specification and class of the vessel, the Secretary of State may make regulations relating to the standards of crew accommodation, which will be recorded in the crew agreement. Crew accommodation includes sleeping rooms, mess halls, sanitary accommodation, hospital accommodation and recreation accommodation provided for seafarers. Regulations made under this section will contain basic standards pertaining to the minimum amount of space permitted by way of sleeping accommodation and the maximum number of people that may use the sleeping accommodation. This section will also stipulate where crew accommodation may be located and the standards relating to the construction, equipment and furnishing of said accommodation. The shipowner also has a duty to maintain and repair said accommodation and ensure that the space is not used for any purposes other than crew accommodation. By recording the requirements

<sup>&</sup>lt;sup>324</sup> The Merchant Shipping Act, 1995 Section 38 (1).

<sup>&</sup>lt;sup>325</sup> The Merchant Shipping Act, 1995 Section 42 (1).

<sup>&</sup>lt;sup>326</sup> The Merchant Shipping Act, 1995 Section 42(2).

<sup>&</sup>lt;sup>327</sup> The Merchant Shipping Act 1995, Section 43(1).

<sup>&</sup>lt;sup>328</sup> The Merchant Shipping Act 1995, Section 43(7).

<sup>329</sup> The Merchant Shipping Act 1995, Section 43(2)(a).

<sup>&</sup>lt;sup>330</sup> The Merchant Shipping Act 1995, Section 42(2)(b).

<sup>331</sup> The Merchant Shipping Act 1995, Section 42(2)(d).

relating to accommodation in the crew agreement, seafarers are protected because they are able to hold shipowners accountable for their living conditions. It ensures seafarers have a healthy and safe environment that is utilised for the sole purpose of accommodation.

#### 4.2.5 Medical care and expenses

Section 45 of the UK Merchant Shipping Act protects a seafarer's rights concerning receiving medical care and expenses. Suppose a seafarer is employed on a UK flagged vessel in foreign waters, and requires immediate medical or surgical treatment (including dental and optical care), the shipowner will be responsible for the reasonable costs of such medical treatment.<sup>332</sup> In addition, if a seafarer employed on a UK flagged vessel dies and has to be buried or cremated outside of the UK, the shipowner will be responsible for the costs thereof.<sup>333</sup> When a doctor is not present on board the vessel and a seafarer requires medical attention, the master or any person appointed for this purpose, is responsible for ensuring that all arrangements securing medical attention are made.<sup>334</sup> This ensures that seafarers have access to adequate medical care while employed outside of the UK, the reasonable costs of which will be covered by the shipowner. Thus, ensuring that seafarers are not left vulnerable without adequate medical care, and medical bills that they may not be in a position to pay.

### 4.2.6 Repatriation and relief

The employer must make the necessary arrangements and provisions for the return of any seafarer employed on a UK flagged vessel who is left behind in a country outside the UK or taken to another country on being shipwrecked. <sup>335</sup> The employer must also provide for the seafarers' relief and maintenance until they return to their homeport or country. <sup>336</sup> The provisions to be made by the employer include the repayment of expenses incurred in bringing the shipwrecked seafarer ashore and maintaining him until he is ashore, as well as the payment of burial or cremation expenses should the seafarer die before he is brought ashore. <sup>337</sup> If the seafarer incurs

<sup>&</sup>lt;sup>332</sup> The Merchant Shipping Act 1995, Section 45 (1).

<sup>&</sup>lt;sup>333</sup> The Merchant Shipping Act 1995, Section 45(2).

<sup>&</sup>lt;sup>334</sup> The Merchant Shipping Act 1995, Section 53.

<sup>&</sup>lt;sup>335</sup> The Merchant Shipping Act 1995, Section 73.

<sup>&</sup>lt;sup>336</sup> The Merchant Shipping Act 1995, Section 73(1).

<sup>&</sup>lt;sup>337</sup> The Merchant Shipping Act 1995, Section 73(2).

any expenses, he is entitled to recover these expenses from the employer unless the employer can prove that the expenses were to be borne by the seafarer under the employment contract.<sup>338</sup>

#### 4.3 South Africa

On 20 August 2013, the MLC, which applies internationally, came into force. The MLC is described as the "Bill of Rights for Seafarers" and is a fundamental international instrument for the protection of seafarers.<sup>339</sup> On 20 June 2013, South Africa ratified the MLC; thus bearing the obligation to bring the MLC into force in its domestic legislation within twelve months from the date on which its ratification was registered.<sup>340</sup> *The Merchant Shipping Amendment Act 12 of 2015* gave domestic effect to the MLC in June 2016.<sup>341</sup> Under article 22 of the ILO Constitution, South Africa must report to the ILO, the measures it has implemented in order to give effect to the MLC.
<sup>342</sup> The requirement of this report is part of the ILO's supervisory mechanism, which aims to ensure that full effect is given to the MLC by member states.

## 4.3.1 Crew Agreements

The master or owner of every vessel flagged in South Africa must enter into an agreement or contract with every seafarer engaged in the service of that vessel.<sup>343</sup> The agreement concluded with seafarers must include the nature and duration of the intended voyage or period of engagement, or the maximum period of the voyage or period of employment.<sup>344</sup> The crew agreement must also contain information relating to any places or parts of the world to which the voyage/engagement does not

<sup>&</sup>lt;sup>338</sup> The Merchant Shipping Act 1995, Section 75(2)(b).

<sup>339</sup> Staniland H. "Admiralty Law" (2016) *Annual Survey of South African Law*. Jutastat. <a href="http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm</a> (Accessed 3 March 2021).

<sup>340</sup> Staniland H. "Admiralty Law" (2016) *Annual Survey of South African Law*. Jutastat. <a href="http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm</a> (Accessed 3 March

<sup>341</sup> Staniland H. "Admiralty Law" (2016) *Annual Survey of South African Law*. Jutastat. <a href="http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm</a> (Accessed 3 March 2021).

<sup>&</sup>lt;sup>342</sup> Staniland H. "Admiralty Law" (2016) *Annual Survey of South African Law*. Jutastat. http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm (Accessed 3 March 2021).

<sup>&</sup>lt;sup>343</sup> The Merchant Shipping Act 57 of 1951, Section 103(1).

<sup>344</sup> The Merchant Shipping Act 57 of 1951, Section 103 (3)(a).

extend.345 The crew agreement must also include details as to the number and description of the crew,<sup>346</sup> the time at which the seafarer must be present onboard the vessel when calling at port and the time the seafarer must report for duty,<sup>347</sup> the capacity in which the seafarer is to serve on board the vessel, 348 the amount of wages the seafarer is to receive<sup>349</sup> and the scale of provisions that the seafarer will be furnished with.350

Crew agreements may not be altered, or any part thereof deleted without the seafarer's consent and a written attestation of a proper officer indicating the consent of the seafarer.<sup>351</sup> At the beginning of every voyage, the master of a South African flagged vessel must have a legible copy of the crew agreement framed and displayed on board the vessel; the agreement must be easily accessible to the crew.<sup>352</sup> Crew agreements ensure that the rights of seafarers are sufficiently protected and ensures that seafarers are not exploited while on board the vessel. The crew agreement contains all pertinent information relating to the duties, working hours and period of the voyage which allows the crew to hold the master or shipowner accountable should there be a breach of contract.

#### 4.3.2 Minimum Age

The employment of children on South African flagged vessels is prohibited. The owner or master of a South African flagged vessel or a vessel engaged in plying between ports in South Africa may not knowingly employ or permit any person under the age of sixteen in any capacity onboard the vessel.<sup>353</sup>

#### 4.3.3 Wages

A seafarer employed on a South African flagged vessel shall receive his wages monthly, and any outstanding wages must be paid to the seafarer when he is

<sup>&</sup>lt;sup>345</sup> See note 81.

<sup>&</sup>lt;sup>346</sup> The Merchant Shipping Act 57 of 1951, Section 103 (3)(c).

<sup>&</sup>lt;sup>347</sup> The Merchant Shipping Act 57 of 1951, Section 103 (3)(d).

<sup>&</sup>lt;sup>348</sup> The Merchant Shipping Act 57 of 1951, Section 103 (3)(e).

<sup>&</sup>lt;sup>349</sup> The Merchant Shipping Act 57 of 1951, Section 103 (3)(f).

<sup>&</sup>lt;sup>350</sup> The Merchant Shipping Act 57 of 1951, Section 103 (3)(g).

<sup>&</sup>lt;sup>351</sup> The Merchant Shipping Act 57 of 1951, Section 108.

<sup>&</sup>lt;sup>352</sup> The Merchant Shipping Act 57 of 1951, Section 109.

<sup>353</sup> The Merchant Shipping Act 57 of 1951, Section 110, as amended by section 8 of The Merchant Shipping Amendment Act 12 of 2015.

discharged from the vessel.<sup>354</sup> The master or owner of the vessel must furnish seafarers with an account of his or her wages on a monthly basis.<sup>355</sup> The master of the vessel must enter various deductions and their amounts as they happen in a record book.<sup>356</sup> Deductions to a seafarer's wages will be prohibited unless they are included in the account delivered to the seafarer.<sup>357</sup>

The seafarers' rights to wages and provisions while employed on a South African vessel will be deemed to begin at the time work commences or according to the date specified in the crew agreement.<sup>358</sup>

Should the services of a seafarer be terminated before the date contemplated in the crew agreement, due to a seafarer being left behind because of incapacity, illness or injury, or because of the wreck or loss of the vessel, then the seafarer will be entitled to his/her basic wages until s/he can safely return to port.<sup>359</sup>

This ensures that seafarers on board South African vessels are paid their wages monthly and will not be subjected to long periods without having received their wages. In addition, recording all deductions, protects seafarers from having unauthorised amounts deducted from their wages without their knowledge. Not only will seafarers be aware of how many deductions were made, but will also be able to keep track of what the deductions were for. This allows seafarers who belong to unions or pension funds to ensure that these deductions on their wages are paid towards their pension fund and/or union and that the correct amounts are deducted.

#### 4.3.4 Repatriation and relief

When the seafarer's service on a SA flagged vessel is terminated without the seafarer's consent or before the termination date in the agreement, the owner must make provision for the maintenance of the seafarer and for the return of that seafarer to the proper return port.<sup>360</sup> If the owner or master fails to maintain and return the

<sup>&</sup>lt;sup>354</sup> The Merchant Shipping Act 57 of 1951, Section 120.

<sup>&</sup>lt;sup>355</sup> The Merchant Shipping Act 57 of 1951, Section 121 (1), as amended by section 14 of The Merchant Shipping Amendment Act 12 of 2015.

<sup>&</sup>lt;sup>356</sup> The Merchant Shipping Act 57 of 1951, Section 121(4).

<sup>&</sup>lt;sup>357</sup> The Merchant Shipping Act 57 of 1951, Section 121(3).

<sup>&</sup>lt;sup>358</sup> The Merchant Shipping Act 57 of 1951, Section 133.

<sup>&</sup>lt;sup>359</sup> The Merchant Shipping Act 57 of 1951, Section 140(1).

<sup>&</sup>lt;sup>360</sup> The Merchant Shipping Act 57 of 1951, Section 114(1), as amended by Section 11 of The Merchant Shipping Amendment Act 12 of 2015.

seafarer to the appropriate return port, the expenses borne by the seafarer will be recoverable as wages due to him.<sup>361</sup>

In terms of Section 154(1)(a) of the Merchant Shipping Act, a distressed seafarer is described as a seafarer who has been discharged or left behind from a shipwrecked vessel and is in distress in a place outside of South Africa. Repatriation expenses are incurred by the shipowner to return a seafarer to the proper return port and provide the seafarer with adequate clothing and maintenance until he arrives at the said port. In the case of a shipwrecked seafarer, it refers to the expenses incurred by the shipowner in transporting and maintaining him until he arrives at the return port.

If a seafarer employed on a wrecked or lost vessel sustains a loss of any or all property, the vessel owner will be liable to compensate the seafarer for the loss of their property.<sup>365</sup>

## 4.3.5 Medical Care

The owner or master of a South African flagged vessel must ensure an adequate supply of medication and appliances to treat and prevent illnesses and injuries likely to occur at sea onboard the vessel.<sup>366</sup> The shipowner will be responsible for the expenses of all medical attention and treatment given to a seafarer whilst on board the vessel.<sup>367</sup>

Where a seafarer onboard an SA flagged vessel is injured or suffers from any illness, resulting in the seafarer being discharged or left behind at a foreign port, the shipowner will be liable for providing the necessary medical care, treatment, medication and the expenses of maintaining the seafarer in a manner appropriate to their rank until the

<sup>&</sup>lt;sup>361</sup> The Merchant Shipping Act 57 of 1951, Section 114(2).

<sup>&</sup>lt;sup>362</sup> The Merchant Shipping Act 57 of 1951, Section 154(1)(a).

<sup>&</sup>lt;sup>363</sup> The Merchant Shipping Act 57 of 1951, Section 154(1)(b).

<sup>&</sup>lt;sup>364</sup> The Merchant Shipping Act 57 of 1951, Section 154(1)(b).

<sup>&</sup>lt;sup>365</sup> The Merchant Shipping Act 57 of 1951, Section 140(3).

<sup>&</sup>lt;sup>366</sup> The Merchant Shipping Act 57 of 1951, Section 167.

<sup>&</sup>lt;sup>367</sup> The Merchant Shipping Act 57 of 1951, Section 169 as amended by section 20(3) of The Merchant Shipping Amendment Act 12 of 2015.

seafarer is cured, dies or arrives at the proper return port.<sup>368</sup> Where a seafarer dies, the shipowner will be liable for the expense of the seafarers' burial or cremation.<sup>369</sup>

Where a seafarer is temporarily removed from the vessel at a foreign port to prevent infection or for the convenience of the ship and later returns to duty, the shipowner will be liable for the expenses of the removal, medical treatment and maintenance of the seafarer while they were away from the vessel.<sup>370</sup>

Suppose any other person pays any expenses that the shipowner should have paid regarding illness or injury of a seafarer. In that case, those expenses must be repaid by the shipowner, and if not paid, will constitute a debt that the shipowner is liable for.<sup>371</sup>

These regulations, ensure that South African seafarers have access to adequate medical care while onboard a vessel and when calling at foreign ports. It ensures that a South African seafarer who is injured or ill will not be without access to medical care due to the seafarer being unable to afford medical bills. In addition, a seafarer who takes ill and remains outside South Africa, will not be left in a vulnerable position because the shipowner is responsible for ensuring that the seafarer returns to his/her home port. It is only upon the safe return of the seafarer to his/her home port that the shipowner's duty is discharged.

## 4.3.6 The Merchant Shipping Bill 2020

In 2017, South Africa aimed to facilitate the growth and development of its maritime transport system as a means to improve socio-economic development in the country, whilst also making a substantial contribution to international trade, as a result of which the Comprehensive Maritime Transport Policy (CMTP) was approved.<sup>372</sup> This led to the drafting of the Merchant Shipping Bill, 2020 (the Bill).<sup>373</sup> The purpose of the Bill is

<sup>372</sup> I Surian 'The South African Merchant Shipping Bill, 2020' *LexisNexis* 02 July 2020 Available at: <a href="https://www.lexisnexis.co.za/lexis-digest/legal/maritime-law/the-south-african-merchant-shipping-bill,-2020">https://www.lexisnexis.co.za/lexis-digest/legal/maritime-law/the-south-african-merchant-shipping-bill,-2020</a> (accessed on 10 November 2021).

<sup>&</sup>lt;sup>368</sup> The Merchant Shipping Act 57 of 1951, Section 169 as amended by section 20(1) of The Merchant Shipping Amendment Act 12 of 2015.

<sup>&</sup>lt;sup>369</sup> The Merchant Shipping Act 57 of 1951, Section 169 as amended by section 20(1) of The Merchant Shipping Amendment Act 12 of 2015.

<sup>&</sup>lt;sup>370</sup> The Merchant Shipping Act 57 of 1951, Section 169 as amended by section 20(2) of The Merchant Shipping Amendment Act 12 of 2015.

<sup>&</sup>lt;sup>371</sup> The Merchant Shipping Act 57 of 1951, Section 170.

<sup>&</sup>lt;sup>373</sup> Merchant Shipping Bill, 2020, published for comment in GN 148 in GG 43073 6 March 2020.

to allow South Africa to align itself with international standards and ratify various IMO Conventions.<sup>374</sup> The Bill aligns with the shipping provisions of the CMTP. The Bill was presented and discussed by stakeholders in meetings convened nationally by the Department of Transport during 2018 and 2019. The Bill was tabled and approved by the Directors General Cluster of International Cooperation, Trade and Security Cluster and the Economic Cluster.<sup>375</sup>

Currently the rights of seafarers are protected under the Merchant Shipping Act 51 of 1951, that is until the new Bill is passed into law. The Merchant Shipping Bill is primarily aimed at reviving the maritime transport sector and enhancing its contribution to the growth and radical transformation of the South African economy.<sup>376</sup>

It is intended that the Bill will a have a major impact on the rights of seafarers, greatly improving their working and living conditions. The Bill was drafted with the following objectives:<sup>377</sup>

- i. To make provision for the powers and duties of the Minister and the South African Maritime Safety Authority in the administration of the Bill,
- ii. To make provision for the licensing and registration of vessels in South Africa;
- iii. To make provision for the application of Labour Laws to extend to seafarers, their conditions of employment and their health and well-being whilst on board vessels;
- iv. To make provision for the safety of life at sea and to establish inspection and enforcement mechanisms including those for marine casualties and crimes committed on ships;
- v. to make provision for the regulation of marine traffic, legal proceedings and jurisdictional matters; and
- vi. to recognise and incorporate international conventions to which South Africa is bound in terms of the provisions of the Constitution.

-

<sup>&</sup>lt;sup>374</sup> Ibid 627.

<sup>&</sup>lt;sup>375</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

<sup>376</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill,

<sup>2020&#</sup>x27; *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

377 | Ibid.

Once approved, the Bill aims to effectively amend and repeal several pieces of maritime related legislation, including the following:<sup>378</sup>

- i. repeal the Merchant Shipping Act 57 of 1951;
- ii. repeal the Marine Traffic Act 2 of 1981;
- iii. repeal the Ship Registration Act 58 of 1998;
- iv. amend s1 of the National Ports Act 12 of 2005; and
- v. repeal Annexure 1 of the Ports Rules GN255 GG31986/6-3-2009.

Chapter 4 of the Bill is an integral chapter, aimed at improving the rights and working conditions of South African seafarers. This Chapter of the Bill incorporates existing domestic labour laws, making these applicable to seafarers. This means that in addition to the rights afforded to seafarers under the current Merchant Shipping Act and the Maritime Labour Convention, seafarers will have the added protection of the Basic Conditions of Employment Act, the Labour Relations Act, the Occupational Health and Safety Act and the Compensation for Occupational Injuries and Diseases Act.<sup>379</sup> Chapter 4 of the Bill makes further provisions for the conditions of employment of seafarers, their health and general well-being; 181 rights of distressed seafarers; the rights of repatriation of seafarers; and sanctions for misconduct and offences by the Master.

It is, however, pertinent to note that The Merchant Shipping Bill 2020 is a draft bill and there is no indication of when, or if it will even come into operation. Because the Bill is still in its draft form, there is limited literature available on the Bill and the implications of its coming into operation. Therefore, an in-depth analysis of the Bill, particularly Chapter 4 of the Bill and its implications in effectively promoting the rights of seafarers, is outside the scope of this study.

<sup>&</sup>lt;sup>378</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

<sup>&</sup>lt;sup>379</sup> Sections 73-76 of the Merchant Shipping Bill, 2020, published for comment in GN 148 in GG 43073 6 March 2020.

<sup>&</sup>lt;sup>380</sup> Sections 87-110 of the Merchant Shipping Bill, 2020, published for comment in GN 148 in GG 43073 6 March 2020.

<sup>&</sup>lt;sup>381</sup> Sections 119-133 of the Merchant Shipping Bill, 2020, published for comment in GN 148 in GG 43073 6 March 2020.

The Bill is a vast document of more than 400 pages, and is likely to draw significant comment from a wide range of industry sectors ahead of the promulgation process. Attendance was low at the sessions in Johannesburg and Cape Town where the bill was tabled for comment and discussion. Stakeholders are of the opinion that the Bill should undergo significant scrutiny and are sceptical of the Bill being passed in its present form. Bill being

The current Merchant Shipping Act 51 of 1951 was amended in 2015 to ensure that it is in line with the Maritime Labour Convention, the new Bill once passed into law, would go even further in protecting the rights and interests of seafarers. Despite this huge step towards promoting the rights of seafarers, concerns have been raised regarding government's claims of boosting future economic growth through an inclusive maritime sector. Industry stakeholders warn that drafting maritime legislation with narrow domestic interests in mind, could have unforeseen negative consequences on several industries.<sup>384</sup>

The Bill states that no ship other than a South African owned ship is permitted to participate in coastwise traffic for the transportation of goods between ports in SA. It further states that those ships participating in coastal shipping will need to apply for a license to do so.<sup>385</sup> If the Bill comes into effect in its current form, foreign vessels will be required to choose only one port of call in SA to offload all their cargo, irrespective of the final destination of the goods within the country.<sup>386</sup> The South African government believes that this will assist with creating opportunities for local ship owners.

2

<sup>382</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

383 N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

385 N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

386 N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

Effectively, this means that cargo cannot be transhipped via a coastal shipping network and may end up being transported via trucks and rail for onward moving. The South African government asserts that more studies are required to gage the implications for ports and other domestic transport logistics, and has emphasised that it would take several years for a cabotage regime to be fully implemented in South Africa.<sup>387</sup>

South Africa has never had a strong local ship owning industry. The South African maritime industry is heavily reliant on international trade with international vessels regularly calling into South African ports, thus South Africa has some of the busiest and well-resourced ports in Africa. Concerns are raised that the proposed changes in the Draft Bill will create unnecessary hurdles, hindering international trade, thereby negatively impacting the economy and substantially reducing port traffic and income in South Africa. <sup>388</sup>

Despite the Bill providing seafarers with extensive rights and protections there are concerns regarding the practicalities of seafarers outside of South Africa accessing the Commission for Conciliation, Mediation and Arbitration (CCMA) and the Labour Court, as well as their legal right to strike as specified in the draft Bill. The Bill does not provide any guidelines as to how and when these rights are to be implemented, creating room for abuse that may have serious economic implications for shipowners and the industry at large.

Should seafarers decide to go on strike when a vessel calls at a foreign port, this may result in the vessel becoming stuck in port for indefinite periods of time. This scenario would not only have huge financial implications on the shipowner who will become liable for costs such as demurrage, but will also create logistical problems in port due to a berth being occupied for an indefinite period of time. This scenario would enable

-

industry-with-the-globe-through-the-merchant-shipping-bill-2020/ (Accessed on 05 December 2021)

<sup>&</sup>lt;sup>387</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

<sup>388</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-africa-al

crew to force shipowners into meeting their demands because of the threat of going on strike while calling at a foreign port. This will in effect negate any form of negotiation or dispute resolution between shipowners and crew when disputes arise. While the right to strike is an important one, due to the difficulty in effectively regulating the right, it has not been included in the MLC and it is not a right afforded to American or English seafarers under their applicable laws. Should the Bill be passed, the right to strike will be a right exclusively afforded to South African seafarers

Due to the broad ambit of the Bill and the implications it will have by repealing several preceding. Acts, coupled with the vague guidelines regarding seafarers' rights contained in the Bill, stakeholders in the South African maritime industry will need to mobilise effectively to ensure that the final draft of the Bill meets the needs of the South African maritime industry as a whole. Stakeholders in the maritime industry will need to ensure their concerns regarding the implementation of a seafarer's right to strike and the cabotage regime are adequately addressed to ensure that the final draft of the Bill not only meets the current needs of the South African maritime industry and economy, but also aligns with the needs and vision of the international maritime industry.

\_

<sup>&</sup>lt;sup>390</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

<sup>391</sup> N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

#### **CHAPTER FIVE: CONCLUSION**

The preceding chapters of this study attempt to provide a clearer picture of the nature of MLC Convention and what it aims to achieve. It examines the position of seafarers prior to the implementation of the MLC as well as the current position of seafarers within the respective jurisdictions.

This chapter will conclude on the findings of the previous chapters and address the following research questions to determine if the MLC has been effective in protecting the rights of seafarers:

- 1. To what extent were seafarers' rights protected when they were injured during their employment and at what point did a shipowner's liability end when a seafarer was injured prior to the coming into operation of the MLC?
- 2. To what extent are seafarers' rights currently protected when they are injured during their employment and at what point does a shipowner's liability end once a seafarer has been injured after the implementation of the MLC?
- 3. Has the implementation of the MLC been successful in ensuring the protection of seafarer's rights when they are injured, and has the MLC clearly identified the end of a shipowner's liability?
- 4. Has the MLC been successful in ensuring seafarers are treated fairly and are at an equal playing field and has it created uniformity with regard to their treatment?

#### 5.1 The Maritime Labour Convention

The MLC is a comprehensive legal instrument that contains a vast set of international standards contained in preceding labour instruments. It consolidates all but four of the preceding maritime labour conventions into one convention.<sup>392</sup> The MLC was created to address the challenges brought about by the multitude of labour conventions that countries had to ratify and implement previously, making it difficult to create a cohesive industry standard with regard to the rights of seafarers.<sup>393</sup> In addition, many of these

<sup>&</sup>lt;sup>392</sup> See note 47.

<sup>&</sup>lt;sup>393</sup> See note 64.

conventions have become outdated and do not sufficiently provide for the current living and working conditions on board modern vessels.<sup>394</sup>

Additionally, many of the existing conventions have achieved very low levels of ratification, often making them ineffective. Furthermore, there was a dire need to develop practical enforcement and compliance systems that could be used to effectively enforce standards of ship safety and security adopted within the IMO framework.<sup>395</sup> The MLC was designed to be more effective and capable of wider ratification by consolidating and improving on preceding conventions.<sup>396</sup>

The "no more favourable treatment clause" contained in Article V, paragraph 7 of the MLC plays an instrumental role in protecting seafarers' rights. This is because Article V allows for port inspections carried out on vessels calling in ports that have ratified the Convention, to include inspections related to the labour rights of seafarers.<sup>397</sup>

As a result, the MLC has far reaching effects because it applies indirectly even to seafarers' working onboard vessels from non-member states when calling at ports that are member states of the Convention.<sup>398</sup> As of June 2020, ninety-eight (98) States have ratified the MLC, some of which are major shipping nations, therefore, all vessels entering these States will be subject to the labour standards set in the MLC. With such a large number of states having ratified the MLC, it would be beneficial for non-member states to ensure that their national legislation is in line with the minimum requirements set out in the MLC in order to avoid unnecessary delays when calling at Ports of MLC member states.

Despite the MLC being a comprehensive Convention that protects the rights of seafarers, including those on-board vessels that are flagged in non-member states, the MLC is not above criticism. Many of the prescriptive requirements contained in preceding conventions that were problematic in relation to implementation, are set out in Part B of the MLC.<sup>399</sup> Under the MLC, these provisions are set out as guidelines and are therefore not mandatory; as a result, port authorities are not strictly bound by

<sup>&</sup>lt;sup>394</sup> See note 65.

<sup>&</sup>lt;sup>395</sup> See note 67.

<sup>&</sup>lt;sup>396</sup> See note 81.

<sup>&</sup>lt;sup>397</sup> See note 135.

<sup>&</sup>lt;sup>398</sup> See note 137.

<sup>&</sup>lt;sup>399</sup> See note 74.

these guidelines. States are merely required to ensure that their national laws are "substantially equivalent" to these guidelines when implementing them.<sup>400</sup> This leaves the interpretation of what constitutes the standard of "substantially equivalent", to the individual States, thereby creating room for some states to apply a lower standard of the guidelines set out in the MLC.

The MLC has also been criticised for not addressing matters such as the protection of seafarers right to strike and the issuing of visas for shore leave. <sup>401</sup> The crew change crisis of the Covid-19 Pandemic clearly illustrated the vulnerability of seafarers when they are at sea and how important it is for them to have the right to strike. Despite the international "call to action" there were approximately 300 000 seafarers stuck at sea, this clearly indicated the weaknesses in labour regulatory systems. <sup>402</sup> Early criticism of the MLC highlighted the lack of provisions allowing seafarers to strike, and it was only through the Covid-19 Pandemic and the Crew Change Crisis that emphasized that without some enabling provisions, such as the right to strike, other rights protected by the MLC become meaningless. <sup>403</sup>

Nevertheless, the MLC is a great step forward in both consolidating the rights of seafarers and harmonising compliance and inspection procedures to be followed by flag states and port authorities.<sup>404</sup>

### 5.2 The United States of America

Seafarer's rights to recover damages have developed extensively from the initial common law remedies that restricted shipowner liability to maintenance, cure and unearned wages. The Osceola decision was the first of its kind, where a court declared that a seafarer could recover damages for injuries sustained due to an unseaworthy vessel, or the shipowner's failure to provide equipment. The court however held that a seafarer could only claim damages for injuries caused due to

<sup>&</sup>lt;sup>400</sup> See note 151.

<sup>&</sup>lt;sup>401</sup> See note 167.

<sup>&</sup>lt;sup>402</sup> See note 170.

<sup>&</sup>lt;sup>403</sup> See note 171.

<sup>&</sup>lt;sup>404</sup> The Maritime Labour Convention: a review. *Lexology*. Available at <a href="https://www.lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0">https://www.lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0</a> (Accessed on 16

December 2021).

<sup>&</sup>lt;sup>405</sup> See note 167.

<sup>&</sup>lt;sup>406</sup> See note 169.

factors within the direct control of the shipowner, and not for example for injuries caused due to the negligence of the master or members of the crew. 407 The USA Congress then passed the Jones Act, which aimed to extend the rights of seafarers. 408 A significant change brought about by the Jones Act, is that it allowed seafarers to claim damages for injuries sustained due to the negligence of the employees of the shipowner, including the master of the vessel. 409

The courts finding in *The Osceola*, rendered parts of the Seaman's Act of 1915, irrelevant. The President of the International Seaman's Union of America suggested that Section 20 of the Seamen's Act be amended, this amendment was incorporated into Section 33 of the Jones Act of 1920 and provides as follows:

"any seaman who shall suffer personal injury in the course of his employment may, at his election, maintain an action for damages at law, with the right of trial by jury, and in such action all statutes of the United States modifying or extending the common-law right or remedy in cases of personal injury to railway employees shall apply; and in case of the death of any seaman as a result of any such personal injury the personal representative of such seaman may maintain an action for damages at law with the right of trial by jury, and in such action all statutes of the United States conferring or regulating the right of action for death in the case of railway employees shall be applicable. Jurisdiction in such actions shall be under the court of the district in which the defendant employer resides or in which his principal office is located."

In the case of *Romero v International Terminal Operating Co*, the court ultimately held that in a claim made under the Jones Act, unseaworthiness, maintenance and cure could be decided in the same suit provided that the claimant did not receive a double recovery. Therefore, it was decided that the Jones Act together with the doctrine of unseaworthiness, maintenance and cure could be regarded as a single source containing an injured seafarer's full remedies. As a result, under this law an injured seafarer could recover damages in the form of wages for the duration of the voyage, costs for medical care such as medication, hospitalisation and compensation for pain

<sup>&</sup>lt;sup>407</sup> See note 170.

<sup>&</sup>lt;sup>408</sup> See note 172.

<sup>&</sup>lt;sup>409</sup> See note 174.

<sup>&</sup>lt;sup>410</sup> See note 179.

<sup>&</sup>lt;sup>411</sup> See note 199.

<sup>&</sup>lt;sup>412</sup> See note 200.

and suffering, as well as loss of future earnings due to impairment and occasionally for loss of wages.<sup>413</sup>

The USA has been at the forefront of seafarer protection through the implementation of regulations such as the Jones Act. 414 However, the USA has experienced difficulty in protecting the rights of foreign seafarers that enter its ports. 415 The Jones Act only effectively protects American seafarers because USA Port State Control authorities cannot inspect and hold foreign flagged vessels to the standards set in the Jones Act. 416 The most recent ILO Convention that the United States has ratified is Convention no 147, The Merchant Shipping (Minimum Standards) Convention. 417 The Convention aims to improve the conditions of employment on merchant vessels and allow the Port States that are parties to the Convention to take action where necessary to protect the health and safety of seafarers manning merchant vessels calling at those ports. 418 The Convention aimed to introduce minimum labour standards throughout the shipping industry by implementing these standards through practical Port State Control examination and reporting systems. 419 In theory, Convention no 147 was designed to advance seafarers' human and labour rights, however, the Convention has fallen short in terms of its applicability. 420 It is one of several ILO conventions that have been poorly received with only 33 formal ratifications.<sup>421</sup>

The USA has not ratified the MLC and while the Jones Act has been constantly updated and adequately protects the rights of American seafarers, its protection only extends to American seafarers working on USA flagged vessels. At a glance, it would appear that ratifying the MLC would be of no additional benefit to American seafarers, 422 however, it would allow the USA to better protect the rights of all seafarers, regardless of the flag state of the vessel they are employed on. The labour regulations under the Jones Act as well as the general labour laws applicable in the USA are already in line with the provisions of the MLC thus making the ratification

<sup>&</sup>lt;sup>413</sup> See note 201.

<sup>&</sup>lt;sup>414</sup> See note 260.

<sup>&</sup>lt;sup>415</sup> See note 261.

<sup>&</sup>lt;sup>416</sup> See note 266.

<sup>&</sup>lt;sup>417</sup> See note 277.

<sup>&</sup>lt;sup>418</sup> See note 278.

<sup>&</sup>lt;sup>419</sup> See note 279.

<sup>&</sup>lt;sup>420</sup> See note 280.

<sup>&</sup>lt;sup>421</sup> See note 282.

<sup>&</sup>lt;sup>422</sup> See note 267.

thereof relatively easy for the USA.423 Furthermore, by failing to ratify the MLC, USA flagged vessels may be subjected to enhanced scrutiny and inspection when calling at ports of member states of the MLC. 424 While there is limited risk of any significant non-compliance being detected on USA flagged vessels, these in-depth inspections affect seafarers as they have less time available to go ashore and enjoy much needed recreational breaks.425

Ratifying the MLC would allow the USA to further its commitment to addressing labour deficiencies on all flagged vessels calling at its ports, thereby, clamping down on vessels non complaint with basic labour standards provided for in the MLC. 426

Previously in the USA seafarers had very limited rights and remedies available to them the Jones Act was therefore a huge step forward in protecting the interests of seafarers.

The Jones Act and other national legislation as well as ratification of Convention no 147 has ensured that American seafarers' rights on board USA flagged vessels such as enforcing minimum working age, crew agreements, accesses to medical care, minimum wage and working hours, repatriation and maintenance. However, by not ratifying the MLC, the USA has failed to capitalise on the opportunity of further developing global standards protecting seafarers' rights, through its leading Supreme Court decisions of The Osceola<sup>427</sup>, Chelentis v Luckenbach S.S Co,<sup>428</sup> Panama R.R Johnson, 429 Pacific S.S v Peterson 430 and Romero v International Terminal Operating Co.431

<sup>&</sup>lt;sup>423</sup> See note 268.

<sup>&</sup>lt;sup>424</sup> See note 269.

<sup>&</sup>lt;sup>425</sup> See note 270.

<sup>&</sup>lt;sup>426</sup> See note 287.

<sup>&</sup>lt;sup>427</sup> See note 168.

<sup>&</sup>lt;sup>428</sup> See note 177.

<sup>&</sup>lt;sup>429</sup> See note 196.

<sup>&</sup>lt;sup>430</sup> See note 197.

<sup>&</sup>lt;sup>431</sup> See note 198.

#### 5.3 The United Kingdom

It has long been established that the basic rights of a seafarer include maintenance and cure, wages and passage back to port. 432 This formed part of English common law, however, in the UK the rights of seafarers are protected predominantly through statutes and case law. 433 The regulatory measures to protect seafarers developed as early as the 1960's due to concerns regarding the health and safety of seafarers.<sup>434</sup>

One of the earliest codes protecting the rights of seafarers was established in the medieval period. The Rules of Oleron placed a duty on the master of a vessel to ensure the continued care and financial support of ill or injured seafarers.<sup>435</sup> This has become a basic right for seafarers and is contained in several international conventions. 436 Despite the Rules of Oleron placing this duty on seafarers, the British Medical Journal began exposing the poor living conditions and neglect experienced by seafarers.<sup>437</sup> This ultimately led to the development of the Merchant Shipping Act of 1867 which laid the foundation for the provisions used to protect seafarers today.<sup>438</sup>

After World War I seafarers were recognised as the most international group of workers resulting in the development of bodies such as the ILO aimed at enhancing the rights and wellbeing of workers in the maritime sector. 439 As a result, several international conventions were developed in an attempt to protect the wellbeing of seafarers. English law accordingly developed, The Merchant Shipping Act of 1995 improved on and consolidated its previous Merchant Shipping legislation, but it also drew heavily from international conventions.440 The UK implements international regulations, directive and decisions through domestic legislation, such as The Merchant Shipping Act of 1995.441

<sup>&</sup>lt;sup>432</sup> See note 202.

<sup>&</sup>lt;sup>433</sup> See note 203.

<sup>&</sup>lt;sup>434</sup> See note 204.

<sup>&</sup>lt;sup>435</sup> See note 206.

<sup>&</sup>lt;sup>436</sup> See note 207. <sup>437</sup> See note 208.

<sup>&</sup>lt;sup>438</sup> See note 211.

<sup>&</sup>lt;sup>439</sup> See note 216.

<sup>&</sup>lt;sup>440</sup> See note 221.

<sup>&</sup>lt;sup>441</sup> See note 225.

To date the UK has ratified all major international maritime conventions that are implemented through its domestic legislation. While the UK was a member of the European Union, the regulations made by the European Union had a direct or indirect effect on the operation of the UK. After the triggering of Article 50 in March 2017, the UK left the European Union in January 2020 and went into a period of transition that ended on 31 December 2020. After the 31st of December 2020, the regulations made under the European Union would not be applicable to the UK, unless these were expressly implemented into UK law. The UK ratified the MLC in August 2014; therefore, the provisions of the MLC have already been adopted into the UK's domestic legislation. The UK's domestic legislation gives effect to the provisions of the MLC, however, any gaps that may have emerged from it leaving the European Union will need to be addressed. Because the UK has ratified and adopted international conventions through its domestic legislation, it is not yet evident what gaps, if any have emerged through leaving the European Union and it is unlikely that the UK will have to make any significant changes to its legislation.

Prior to the implementation of the MLC, in the UK an injured seafarers' rights and right of recourse against a shipowner included the right to wages that are to be paid monthly. Further, a seafarer is entitled to wages even in instances where a ship has been wrecked or lost. He Merchant Shipping Act 1995 also provides for minimum requirements to ensure satisfactory living conditions for crew. As seafarer is entitled to medical treatment while onboard the vessel and ashore, when outside the UK. In addition where a seafarer died at sea or ashore outside the UK, the shipowner would be liable for burial or cremation costs. Seafarers are also entitled to relief, maintenance and return to the UK at the end of a voyage, or in instances of a shipwreck.

<sup>&</sup>lt;sup>442</sup> See note 293.

<sup>&</sup>lt;sup>443</sup> See note 294.

<sup>&</sup>lt;sup>444</sup> See note 295.

<sup>&</sup>lt;sup>445</sup> See note 296.

<sup>&</sup>lt;sup>446</sup> See note 297.

<sup>&</sup>lt;sup>447</sup> See note 226.

<sup>&</sup>lt;sup>448</sup> See note 227.

<sup>&</sup>lt;sup>449</sup> See note 228.

<sup>&</sup>lt;sup>450</sup> See note 229.

<sup>&</sup>lt;sup>451</sup> See note 230.

<sup>&</sup>lt;sup>452</sup> See note 231.

Once the UK ratified the MLC and implemented the provisions of the MLC it further added to the rights already afforded to seafarers through *The Merchant Shipping Act* 1995. A significant change introduced into its domestic legislation relates to the protection of seafarers caught up in piracy and other criminal acts on board a vessel. The new legislation provides that if a seafarers is held captive on board a vessel, s/he is still entitled to be remunerated by the shipowner and their employment contact would remain in place. While the implementation of the MLC in the UK, has not resulted in major changes to preceding legislation, the MLC has significantly enhanced the protection afforded to English seafarers by ensuring that their crew agreements are given effect to during an act of piracy.

#### 5.4 South Africa

South African admiralty law has a strong link to English law, therefore South African admiralty law can never be considered without reference to the history of law in the UK. 456 The origins of South African maritime law can be traced back to the laws of the first European settlers in the Cape in 1652. 457 South African common law is a combination of Roman-Dutch and English law; the British established the court system and British judges presided over matters. 458

Prior to enacting AJRA, English statutes relating to admiralty matters as of 1891 were applicable in South Africa, therefore South African courts were to apply English admiralty law and precedents. The application of s6 of AJRA has resulted in wage claims in South Africa being subject to English law as it were in 1893; however, other claims relating to employment would be subject to South African Roman-Dutch law. Therefore, a seafarers right to wages would have been protected under English law and applied in South Africa. Any other rights and claims of a seafarer would be protected under South African law through the *Merchant Shipping Act*.

<sup>&</sup>lt;sup>453</sup> See note 292.

<sup>&</sup>lt;sup>454</sup> See note 312.

<sup>&</sup>lt;sup>455</sup> See note 313.

<sup>&</sup>lt;sup>456</sup> See note 232.

<sup>&</sup>lt;sup>457</sup> See note 235.

<sup>&</sup>lt;sup>458</sup> See note 238.

<sup>&</sup>lt;sup>459</sup> See note 242.

<sup>&</sup>lt;sup>460</sup> See note 249.

<sup>&</sup>lt;sup>461</sup> See note 250.

On the 20<sup>th</sup> of June 2013, South Africa ratified the MLC, as a result of which South Africa has an obligation to ensure that it gives effect to the provisions of the MLC through its domestic legislation.<sup>462</sup> *The Merchant Shipping Amendment Act 12 of 2015,* gave domestic effect to the MLC in June 2016.<sup>463</sup>

Prior to the implementation of the MLC, South African seafarers had the same rights and remedies as English seafarers. They were entitled to the full payment of their wages, 464 repatriation expenses, 465 maintenance, the provision of necessaries, accommodation 466 and proper medical care. 467

Since the implementation of *The Merchant Shipping Amendment Act 12 of 2015*, there has not been a significant change to the rights of seafarers, however, through the implementation of the MLC, the existing rights of seafarers have been further enhanced and developed. This is evident through the additional requirements that have been added to crew agreements and the specifications that must be followed with regard to crew accommodation. While some of these additional requirements may not be drastic in nature, they have a significant effect in improving the health, safety and overall wellbeing of the crew while onboard vessels.

Should *The Merchant Shipping Bill 2020*, come into effect this would have a significant effect on the rights and remedies available to seafarers. The application of the Bill would extend the provisions of the Labour Relations Act, the Occupational Health and Safety Act and the Compensation for Occupational Injuries and Diseases Act, to seafarers. This would place South African seafarers in a more favourable position than most of their international counterparts.

While the Bill may be a huge step towards better protecting the rights of South African seafarers, certain provisions of the Bill raises great concerns. Due to the international nature of the shipping industry, provision for seafarers to access the CCMA and Labour Court as well as enforcing the legal right to strike, will introduce significant

<sup>&</sup>lt;sup>462</sup> See note 332.

<sup>&</sup>lt;sup>463</sup> See note 334.

<sup>&</sup>lt;sup>464</sup> See note 251.

<sup>&</sup>lt;sup>465</sup> See note 254.

<sup>&</sup>lt;sup>466</sup> See note 257.

<sup>&</sup>lt;sup>467</sup> See note 259.

<sup>&</sup>lt;sup>468</sup> See note 370.

<sup>&</sup>lt;sup>469</sup> See note 373.

challenges to shipowners and the industry at large.<sup>470</sup> Affording these additional rights and remedies to seafarers would inevitably result in higher expenses and liability of shipowners registering their vessels in South Africa, thus further deterring ship registration in this country.

A further concern to the maritime industry is that should the Bill come into effect only South African owned vessels will be permitted to participate in coastwise traffic for transporting goods between South African ports. <sup>471</sup> Effectively, this means that foreign vessels will be required to call at only one port in South Africa. <sup>472</sup> The South African government believes that this will assist with creating opportunities for local shipbuilders, similarly to the American shipbuilding industry that operates in the same way.

While implementing a cabotage regime would appear to strengthen the South African economy and create opportunities for South African entrepreneurs, this regime could inevitably deter international markets and destroy local ports. Outh Africa has some of the busiest ports in Africa, if the cabotage regime is implemented most cargo will need to be transported to their final destination either by rail or road. Currently South Africa does not have the infrastructure or the economy to effectively facilitate the transportation of cargo inland. Consequently, this could result in long delays in cargo arriving at their intended destination. These added delays and costs associated with the transport of goods via truck and rail could create several market related hurdles, no longer making South African ports lucrative to the international shipping market. For the cabotage regime to be successful the South African government will need to do more studies to determine the implications on logistics and the upgrades that will need to be made to our current infrastructure.

#### 5.5 Final remarks and observations

When examining the impact of the MLC on the rights of seafarers in the three jurisdictions discussed, it is evident that despite the USA not being a member state, seafarers in all three jurisdictions are protected and have fairly equal rights and

<sup>&</sup>lt;sup>470</sup> See note 382.

<sup>&</sup>lt;sup>471</sup> See note 385.

<sup>&</sup>lt;sup>472</sup> See note 386.

<sup>&</sup>lt;sup>473</sup> See note 388.

remedies available to them. However, when looking at the UK and South Africa, it is evident that while the implementation of the MLC has not created drastic changes to their national legislation, it has provided more clarity on the extent of the rights and remedies available to seafarers. The MLC expressly provides for the standards and dimensions of implementation in member states, as well as what constitutes adequate crew accommodation. The MLC further sheds clarity on the extent of medical care and services that should be provided to seafarers both while onboard the vessel and while ashore. It expressly states the requirements of the contents, accessibility and implementation of crew agreements. The MLC also provides clarity on repatriation and the extent of liability of shipowners when seafarers are stranded, or shipwrecked. These are just a few examples of how the MLC ensures that seafarers are equally protected in its member states.

While the MLC has not had a drastic effect in the abovementioned jurisdictions, it is pertinent to note that domestic legislation protecting seafarers, even prior to the implementation of the MLC in these jurisdictions, already extensively protected seafarers employed on board vessels registered in the UK and SA. The MLC will however have a substantial effect on the ability of port authorities in member states to inspect and take action against owners of substandard vessels due to the "no more favourable treatment" clause. This is because Article V of the MLC will allow member states to conduct inspections relating to labour standards and conditions when these vessels call at their ports. Through these inspections, member states will be in a position to hold these substandard vessels accountable and ensure that they adhere to the provisions in the MLC thus, effectively protecting seafarers globally, and ensuring that they are treated equally.

As discussed above, the MLC is not above criticism. The crew change crisis during the Covid-19 pandemic, highlights the shortfalls in the MLC and the potential this creates to leave seafarers vulnerable. Nevertheless, the MLC is still a great step forward in the consolidation and harmonisation of compliance and inspection procedures followed by flag states and port authorities. Furthermore, the inclusion of Article V, ensures that it is better able to protect the interest of seafarers globally, in comparison to preceding conventions.

## **BIBLIOGRAPHY**

### A. TABLE OF STATUTES

## **INTERNATIONAL CONVENTIONS AND CODES**

Final Report: Joint Maritime Commission (29th Session), Geneva, 22-26 January 2001. 2001. Geneva: ILO.

International Convention for the Safety of Life at Sea (SOLAS) 1974

The Health Protection and Medical Care (Seafarers) Convention, 1987

The Maritime Labour Convention, 2006.

The Merchant Shipping (Minimum Standards) Convention, No 147

The Rules of Oleron, c. 1266. Available at <a href="https://www.admiraltylawguide.com/documents/oleron.html">www.admiraltylawguide.com/documents/oleron.html</a> (accessed 11 May 2020).

The Seamen's Welfare in Ports Recommendation No 48.

Repatriation of Seafarers Convention (Revised) 1987 (No. 166)

Seafarers' Hours of Work and the Manning of Ships Convention 1996 (no. 180)

Shipowners' Liability (Sick and Injured Seamen) Convention, 1936

Unemployment Indemnity (Shipwreck) Convention, 1920 (No. 8)

United Nations Convention on the Sea Law, 1982.

#### SOUTH AFRICAN LEGISLATION

Admiralty Jurisdiction and Regulation Act 105 of 1983.

Merchant Shipping Act 57 of 1951.

Merchant Shipping Amendment Act 12 of 2015.

The Merchant Shipping Bill, 2020.

### **ENGLISH LEGISLATION**

Merchant Shipping Act Commencement no 1 of 1970.

Merchant Shipping (liability of shipowners and others) Act 1958.

Merchant Shipping and Maritime Security Act Of 1997

## **USA LEGISLATION**

The Jones Act of 1920.

The Merchant Marine Act Of 1970.

## **B. TABLE OF CASES**

## SOUTH AFRICAN CASE LAW

Transol Bunker v MV Andrico Unity and Others 1989 (4) SA 325 (A).

#### USA CASE LAW

Chelentis v Luckenbach S.S Co, 247 U.S 372 (1918).

Pacific S.S v Peterson 264 US 375, AMC 1932 (1928).

Panama R.R Johnson 264 US 375, AMC 551 (1924).

Romero v International Terminal Operating Co 358 US 354, AMC 832 (1959).\

## C. <u>SECONDARY SOURCES</u>

## **TEXTBOOKS**

DC Jackson. Enforcement of Maritime Claims. 4th ed. (2005)

Fagan, Roman-Dutch Law in its South African Historical Context in Southern Cross-Civil law and Common Law in South Africa (1996).

L. Christodoulou-Varotsi and D.A Pentsov, *Maritime Work Law Fundamentals:* Responsible Shipowners, Reliable Seafarers. (2008).

T Carter. Merchant Seamen's Health, 1860-1960: Medicine, Technology, Shipowners and the State in Britain. (2014).

TJ Schoenbaum. Admiralty and Maritime Law. 5th Hornbook ed. (2001)

## **THESES**

A Kennerley. *British Seamen's Missions and Sailors' Homes 1815-1970: voluntary welfare provision for serving seafarers.* PhD thesis. University of Plymouth, (1989).

#### **JOURNAL ARTICLES**

Admiralty. Extent of Liability to Injured Seaman. (1921) 30 (8) The Yale Law Journal

Fitz- Henry Smith, Jr "Liability in the Admiralty for Injuries to Seamen." (1906) 19 (1) *Harvard Law Review.* 

George Eddings, Andrew Chamberlain, Holly Colaco and Isabel Philips. "The Shipping Law Review: United Kingdom- England and Wales." (2020)

George W Stumberg, 'The Jones Act. Remedies of Seamen' (1956) 17 *Ohio State Law Journal*.

Louis J Gusmano, 'Seamen's Rights to Recover for injury against Either Shipowner or Charterer' (1980-1981) 55 Tul L Rev 1029.

Max D Kossoris and Joseph Zisman, 'Workmen's Compensation for Seamen' (1946) 62 Monthly Labour Review.

Origins of Admiralty Jurisdiction & Shipping Law in South Africa: The Anglo -Dutch juxtaposition- common law in a civilian system. (2009) *Juta and Company (Pty) Ltd.*Available

<a href="http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/7/8?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/7/8?f=templates\$fn=default.htm</a>.

(Accessed 10 June 2020).

P Link, "One Small Step for the United States, May Be One Giant Leap for Seafarers' Rights". (2015) 33 *Hofstra Lab& Emp LJ*.

Robert M Mallano, 'Seamen's Injuries: The Jones Act, Unseaworthiness, and Maintenance and Cure- The Siamese Triplets' (1963) *51 California Law Review.* 

Steven F Friedell, 'The Interplay of the Jones Act and the General Maritime Law' (2017) 48 *Journal of Maritime Law & Commerce*.

#### **WORKING PAPERS & PROJECT REPORTS**

A Mokhtari, M Lafteh, R Hematjoo "Investigation of the Maritime Labor Convention and its legal effects for countries" (2017) *Research Gate*.

Chapter 5, Master and Crew: Jurisdiction and applicable law. (2009) *Juta and Company* (*Pty*) *Ltd.* Available at <a href="http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/ship/2/29/44?f=templates\$fn=default.htm</a>. (Accessed 25 May 2020).

ILO: Adoption of an instrument to consolidate maritime labour standards, Report I (1A).

ILO Convention No. 147, Student Guide (2010) Int'l Labour Org.

I Surian 'The South African Merchant Shipping Bill, 2020' LexisNexis 02 July 2020

Available at: <a href="https://www.lexisnexis.co.za/lexis-digest/legal/maritime-law/the-south-african-merchant-shipping-bill,-2020">https://www.lexisnexis.co.za/lexis-digest/legal/maritime-law/the-south-african-merchant-shipping-bill,-2020</a> (accessed on 10 November 2021).

Staniland H. "Admiralty Law" (2016) *Annual Survey of South African Law*. Jutastat. Available at <a href="http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm">http://jutastat.juta.co.za/nxt/gateway.dll/jelj/assa/3/39/42?f=templates\$fn=default.htm</a> (Accessed 3 March 2021).

US Coast Guard, No 16711.12A, Commandment Instruction (1996). Available at <a href="https://www.uscg.mil/directives/ci/16000-16999.CI\_16711\_12A.pdf">https://www.uscg.mil/directives/ci/16000-16999.CI\_16711\_12A.pdf</a>. (Accessed 8 August 2020).

## **INTERET SOURCES**

A Practical Overview on Enforcement and Defence of Maritime Claims in South Africa: Including Casualties and Carriage of Goods by Sea. (2014) *Shepstone & Wylie Attorneys*. Available at <a href="http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf">http://wylie.co.za/wp-content/uploads/Admiralty-Jurisdiction-and-Practice-in-South-Africa-05.12.2014.pdf</a>. (Accessed 30 May 2020).

Advantages of the Maritime Labour Convention, 2006. <a href="https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS">https://www.ilo.org/global/standards/maritime-labour-convention/what-it-does/WCMS</a> 153450/lang--en/index.htm. (Accessed 17 September 2020).

International Labour Organisation. *Maritime Labour Convention, 2006: Frequently asked questions.* (2012) available at <a href="http://www.ilo.org/mlc">http://www.ilo.org/mlc</a>. (Accessed: 25 May 2020)

International Labour Organisation- available at <a href="https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:80001:0">https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:80001:0</a> accessed on 13 August 2022.

Labour standards: Basic facts on the Maritime Labour Convention 2006. Available at <a href="https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS\_219665/lang--en/index.htm">https://www.iol.org/global/standards/maritime-labour-convention/what-it-does/WCMS\_219665/lang--en/index.htm</a>. (Accessed 22 September 2020).

Michael Finch II, Lawmakers: Repealing Jones Act Would Have a 'Harmful Effect on our Economy," Available at <a href="http://www.al.com/business/index.ssf/2015/01/lawmakers">http://www.al.com/business/index.ssf/2015/01/lawmakers</a> oppose mccain jones act repeal amendment keystone.html. (Accessed 25 June 2020).

N Smuts 'South Africa aligning its maritime industry with the globe through the Merchant Shipping Bill, 2020' *DeRebus* 01 February 2021. Available at <a href="https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/">https://www.derebus.org.za/south-africa-aligning-its-maritime-industry-with-the-globe-through-the-merchant-shipping-bill-2020/</a> (Accessed on 05 December 2021)

Seafarer working and living rights: Maritime Labour Convention- Gov.UK, <a href="https://www.gov.uk/seafarer-working-and-living-rights/maritime-labour-convention">https://www.gov.uk/seafarer-working-and-living-rights/maritime-labour-convention</a> (Accessed 6 March 2020).

The Jones Act- the Foundation of the Merchant Marine. *American Maritime Congress*. (2012). Available at <a href="http://www.americanmaritime.org/about/jonesact.pdf">http://www.americanmaritime.org/about/jonesact.pdf</a>. (Accessed 5 May 2020).

The Maritime Executive. 2 November 2020. <a href="https://www.maritime-executive.com/corporate/uk-law-ensures-continued-employment-rights-for-seafarers-held-captive">https://www.maritime-executive.com/corporate/uk-law-ensures-continued-employment-rights-for-seafarers-held-captive</a> (Accessed 16 February 2021).

The Maritime Labour Convention: a review. *Lexology*. Available at <a href="https://www.lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0">https://www.lexology.com/library/detail.aspx?g=59d3bdd3-03f1-4a58-b6cd-5692ebe845f0</a> (Accessed on 16 December 2021).

'Why the MLC is failing as a regulatory response to the Covid-19 pandemic' *Splash*, 27 January 2021. Available at <a href="https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic/">https://splash247.com/why-the-mlc-is-failing-as-a-regulatory-response-to-the-covid-19-global-pandemic/</a> (Accessed 05 January 2022).



Miss Mikaela Jauden Joseph (215002928) School Of Law Howard College

Dear Miss Mikaela Jauden Joseph,

00006678 Protocol reference number:

Project title: Liability for Injuries to Seafarers- a critical analysis of the extent of shipowner liability under the Maritime Labour Convention.

#### Exemption from Ethics Review

In response to your application received on 19 August 2021 , your school has indicated that the protocol has been granted EXEMPTION FROM ETHICS REVIEW.

Any alteration/s to the exempted research protocol, e.g., Title of the Project, Location of the Study, Research Approach and Methods must be reviewed and approved through an amendment/modification prior to its implementation. The original exemption number must be cited.

For any changes that could result in potential risk, an ethics application including the proposed amendments must be submitted to the relevant UKZN Research Ethics Committee. The original exemption number must be cited.

In case you have further queries, please quote the above reference number.

#### PLEASE NOTE:

Research data should be securely stored in the discipline/department for a period of 5 years.

I take this opportunity of wishing you everything of the best with your study.

Yours sincerely

Mr Simphiwe Peaceful Phungula obo Academic Leader Research

School Of Law

**UKZN Research Ethios Office** Westville Campus, Govan Mbeki Building Postal Address: Private Bag X54001, Durban 4000 Website: http://research.ukzn.ac.za/Research-Ethics/

Founding Compuses: | Edgewood

Howard College

Medical School

Fietermarkburg

Westville

**INSPIRING GREATNESS**